REFORT RESUNES

ED 010 903

FOUR YEARS OF ACADEMIC ACHIEVEMENT AND DISPOSITION OF THE 1961-62 ENTERING FRESHMEN AT THE UNIVERSITY OF MISSISSIFFY COMPARED WITH AMERICAN COLLEGE TEST SCORES.

BY- PHAY, JOHN E. MCDONALD, DOUGLAS

MISSISSIPPI UNIV., UNIVERSITY, BUR. OF INST. RES.

PUB DATE OCT 65

EDRS FRICE MF-\$0.27 HC-\$7.20 180F.

DESCRIPTORS- COLLEGE ENTRANCE EXAMINATIONS, *COLLEGE STUDENTS, *ACADEMIC ACHIEVEMENT, *ADMISSION CRITERIA, GRADE POINT AVERAGE, COLLEGE ADMISSION, DROPOUT RATE, COMPARATIVE STATISTICS, UNIVERSITIES, AMERICAN COLLEGE TEST (ACT)

THE USE OF AMERICAN COLLEGE TEST (ACT) SCORES AS A CRITERION FOR ADMISSION TO INSTITUTIONS OF HIGHER LEARNING PROVIDED PREDICTIVE INFORMATION, BUT ALSO HAD LIMITATIONS. BEGINNING IN 1961 THIS STUDY FOLLOWED 1,062 FRESHMEN OF THE UNIVERSITY OF MISSISSIPPI THROUGH THEIR 4 YEARS OF COLLEGE. THE ACT WAS GIVEN TO ALL INCOMING FRESHMEN SEPARATED INTO THREE GROUPS BASED ON THEIR ACT SCORES. GROUP I CONSISTED OF 102 STUDENTS WITH COMPOSITE STANDARD SCORES OF LESS THAN 15. GROUP. II CONSISTED OF 250 STUDENTS SCORING BETWEEN 15 AND 18. GROUP III CONSISTED OF 709 STUDENTS SCORING HIGHER THAN 18. CORRELATING COMPOSITE SCORES OF GROUPS I AND II WITH CUMULATIVE GRADE POINTS AT THE END OF EACH SCHOOL YEAR RESULTED IN CORRELATIONS VERY CLOSE TO ZERO. SIGNIFICANT CORRELATIONS, HOWEVER, WERE OBTAINED FOR GROUP III. WHEN SCORES WERE CONSIDERED FOR ALL INCOMING FRESHMEN, THE CORRELATION BETWEEN COMPOSITE ACT SCORE AND ACADEMIC ACHIEVEMENT WAS RELATIVELY STRONG AND POSITIVE. TEST LIMITATIONS WERE EVIDENT SINCE DROFOUT RATE AND ACT SCORES WERE NOT HIGHLY CORRELATED. THE HIGHEST DROPOUT RATE WAS AMONG STUDENTS SCORING RELATIVELY WELL ON THE ACT. FURTHERMORE, THE TEST WAS UNACCEPTABLE IN EVALUATING THE ABILITY OF FOREIGN STUDENTS TO SUCCEED IN COLLEGE. OVERALL, THE ACT TEST PROVED TO BE A GOOD INDICATOR OF ACADEMIC SUCCESS OF INCOMING FRESHMEN. (GB)

FOUR YEARS OF ACADEMIC ACHIEVEMENT AND DISPOSITION OF THE 1961-62 ENTERING FRESHMEN AT THE UNIVERSITY OF MISSISSIPPI COMPARED WITH AMERICAN COLLEGE TEST SCORES

John E. Phay and Douglas McDonald

THE UNIVERSITY OF MISSISSIPPI

Bureau of Institutional Research

U.S. DEPARTMENT OF HEALTH, EDUCATION & WELFARE
OFFICE OF EDUCATION

THIS DOCUMENT HAS BEEN REPRODUCED EXACTLY AS RECEIVED FROM THE PERSON OR ORGANIZATION ORIGINATING IT. POINTS OF VIEW OR OPINIONS STATED DO NOT NECESSARILY REPRESENT OFFICIAL OFFICE OF EDUCATION POSITION OR POLICY.

ea god 614



FOUR YEARS OF ACADEMIC ACHIEVEMENT AND DISPOSITION OF THE 1961-62 ENTERING FRESHMEN AT THE UNIVERSITY OF MISSISSIPPI COMPARED WITH AMERICAN COLLEGE TEST SCORES

John E. Phay
Director of Institutional Research
and
Professor of Education

and

Douglas McDonald
Institutional Research Analyst
and
Assistant Professor of Education

U.S. DEPARTMENT OF HEALTH, EDUCATION & WELFARE
OFFICE OF EDUCATION

THIS DOCUMENT HAS BEEN REPRODUCED EXACTLY AS RECEIVED FROM THE PERSON OR ORGANIZATION ORIGINATING IT. POINTS OF VIEW OR OPINIONS STATED DO NOT NECESSARILY REPRESENT OFFICIAL OFFICE OF EDUCATION POSITION OR POLICY.

THE UNIVERSITY OF MISSISSIPPI

Bureau of Institutional Research University, Mississippi October, 1965

FOREWORD

This is the first major study of the Bureau of Institutional Research, established July 1, 1963.

The question of whether or not to require candidates to institutions of higher learning to meet a required score on achievement tests in order to be eligible for admission, has received considerable attention in recent years. This study follows the 1062 freshmen who entered The University of Mississippi in 1961-62 through their four years of college. This class was not governed by the 1962 policy of The University of Mississippi which requires a student to make on the American College Test, a minimum score of 15 for Mississippi students or 19 for out-of-state students for eligibility for admission to The University of Mississippi. Therefore, information is contained in this study of students scoring below the present minimum required for admission as well as for those scoring above. Dispositions of all students, by ACT scores attained, are presented.

It is hoped that the data contained herein will aid in the making of wise and beneficial decisions regarding minimum admission requirements for institutions of higher learning. In analyzing the data for the students involved, the reader is advised that this class went through an unusual period due to circumstances surrounding the entrance of James H. Meredith to The University of Mississippi.

Our thanks are extended to Mrs. Nancy Lotterhos, Secretary to the Director of Institutional Research, for her services in preparing this volume.

John E. Phay Douglas McDonald



TABLE OF CONTENTS

CHAPIT	ER PAGE
I.	INTRODUCTION
	The Problem
	The ACT Test
	Organization of the Study 3
-	Analysis of the 1961-62 Entering Freshmen 5
II.	ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
•	OF LESS THAN 15
	Classification by Schools 16
	Student Achievement
	Summary of Reports
III.	ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
	OF 15, 16, 17, 18
	Classification by Schools
	Student Achievement
	Summary of Reports
IV.	ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
	OF GREATER THAN 18
	Classification by Schools
	Student Achievement
	Summary of Reports
V.	HOLDING POWER OF THE UNIVERSITY OF MISSISSIPPI
	BASED ON COMPOSITE STANDARD SCORES OF THE
	AMERICAN COLLEGE TEST AND ACADEMIC ACHIEVE-
	MENT OF 1961-62 ENTERING FRESHMEN

1	
1	v

CHAPTE	Ŕ																			PAGE
VI.	SUMMARY AN	D C	ON	CL	US.	ION	13	•	•	•		•		•	•	•	•	•	•	154
	Summary		•	•		•		•	•	•	•	•	•	•	æ	•	•	•	•	154
	Conclusi	Ons	·																	150



LIST OF TABLES

TABLE			PAGE
I-1	Freshmen Enrollments by Sex and Residence of 1961-62 Entering Freshmen Based on the American College Test Composite Standard Score	•	6
I-2	Percentiles for ACT English Scores of 1961- 62 Entering Freshmen, The University of Mississippi	•	7
I-3	Percentiles for ACT Mathematics Scores of 1961-62 Entering Freshmen, The University of Mississippi	•	8
I-4	Percentiles for ACT Social Studies Scores of 1961-62 Entering Freshmen, The University of Mississippi	•	9
I-5	Percentiles for ACT Natural Science Scores of 1961-62 Entering Freshmen, The University of Mississippi	•	10
I-6	Percentiles for ACT Composite Scores of 1961- 62 Entering Freshmen, The University of Mississippi	•	11
II-I	Initial Enrollment of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, by Schools, The University of Mississippi, Beginning of First Semester		14
II-2	Enrollment of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, by Schools, The University of Mississippi, Beginning of Second Semester	•	15
II-3	Enrollment of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, by Schools, The University of Mississippi, Beginning of Third Semester	•	18
II-4	Enrollment of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, by Schools, The University of Mississippi, Beginning		
	of Fourth Semester	•	19

TABLE		PAGE
II-5	Enrollment of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, by Schools, The University of Mississippi, Beginning of Fifth Semester	. 21
II-6	Enrollment of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, by Schools, The University of Mississippi, Beginning of Sixth Semester	. 22
II-7	Enrollment of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, by Schools, The University of Mississippi, Beginning of Seventh Semester	2 5
II-8	Enrollment of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, by Schools, The University of Mississippi, Beginning of Eighth Semester	. 26
II-9	Cumulative Grade Point Averages of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, 1961-62	28
II-10	Cumulative Grade Point Averages of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, 1961-63	29
II-11	Cumulative Grade Point Averages of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, 1961-64	30
II-12	Cumulative Grade Point Averages of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, 1961-65	31
II-13	Grade Point Averages of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test Who Attended the 1961 Summer Session, The University of Mississippi	32

TABLE		PAGE
II-14	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, The University of Mississippi, End of First Semester	33
II-15	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, The University of Mississippi, End of Second Semester	34
II-16	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test Who Attended the 1962 Summer Session, The University of Mississippi	35
II-17	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, The University of Mississippi, End of Third Semester	36
II-18	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, The University of Mississippi, End of Fourth Semester	. 37
II - 19	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test Who Attended the 1963 Summer Session, The University of Mississippi	38.
II-20	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, The University of Mississippi, End of Fifth Semester	39
II-21	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, The University of Mississippi, End of Sixth Semester	40
II-22	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard	

TABLE		PAGE
	Scores of Less Than 15 on the American College Test, Who Attended the 1964 Summer Session, The University of Mississippi	e . 41
II-23	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, The University of Mississippi, End of Seventh Semester	e . 42
II-24	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, The University of Mississippi, End of Eighth Semester	e • 43
II-25	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test Who Attended the 1965 Summer Session, The University of Mississippi	e e • 44
II-26	Enrollment of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, by Schools and by Semester, The University of Mississippi, 1961-621964-65	. 46
II-27	Grade Point Averages and Disposition of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, The University of Mississippi, 1961-621964-65	e . 47
II-28	Grade Point Averages and Disposition of 1961-62 Entering Freshmen with Composite Standard Scores of Less Than 15 on the American College Test, by Test Scores and by End of Semester, 1961-621964-65	. 48
III-1	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, by Schools, The University of Mississippi, Beginning of First Semester	. 57
III-2	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Second Semester	. 58

ERIC Full Text Provided by ERIC

TABLE	A	PAGI
III-3	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Third Semester	61
III-4	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Fourth Semester.	62
III-5	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Fifth Semester	65
III-6	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Sixth Semester	66
III-7	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Seventh Semester	69
III-8	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Eighth Semester	7 0
III-9	Cumulative Grade Point Averages of 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, 1961-1962	72
III-10	Cumulative Grade Point Averages of 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, 1961-1963	73
III-11	Cumulative Grade Point Averages of 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, 1961-1964	74

ERIC*

TABLE		PAGE
III-12	Cumulative Grade Point Averages of 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, 1961-1965	75
III-13	Grade Point Averages of 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test Who were Enrolled in the 1961 Summer Session, The University of Mississippi	76
III-14 ,	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, The University of Mississippi, End of First Semester	77
III-15	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, The University of Mississippi, End of Second Semester	78
III-16	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test Who Enrolled in the 1962 Summer Session, The University of Mississippi	7 9
III-17	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, The University of Mississippi, End of Third Semester	80.
III-18	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, The University of Mississippi, End of Fourth Semester	81
[II-19	Grade Point Averages and Disposition of 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test Who Enrolled in the 1963 Summer Session, The University of Mississippi	82

ERIC Profited by ERIC

TABLE		PAGE
III-20	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, The University of Mississippi, End of Fifth Semester	83
III-21	Grade Point Averages and Disposition of 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, The University of Mississippi, End of Sixth Semester	84
III-22	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test Who Enrolled in the 1964 Summer Session, The University of Mississippi	. 85 .
III-23	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, The University of Mississippi, End of Seventh Semester	86
III-24	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, The University of Mississippi, End of Eighth Semester	87
III -2 5	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test Who Enrolled in the 1965 Summer Session	88
III-26	Enrollment of 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, The University of Mississippi, 1961-621964-65.	90
III-27	Grade Point Averages and Disposition of 1001-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, The University of Mississippi, 1961-621964-65	91

TABLE	•	PAGE
III-28	Grade Point Averages and Disposition of 1961-62 Entering Freshmen with Composite Standard Scores of 15, 16, 17, 18 on the American College Test, by Test Scores and by End of Semester, 1961-621964-65	 .92
IV-1	Initial Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, by Schools, The University of Mississippi, Beginning of First Semester	99
IV-2	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Second Semester	100
IV-3	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Third Semester	103
IV-4	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Fourth Semester.	104
IV-5	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Fifth Semester	107
IV-6	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Sixth Semester	108
IV-7	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Seventh Semester	112

TABLE	•	PAGE
IV-8	Enrollment of the 1961-62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, by Schools, The University of Mississippi, Beginning of Eighth Semester	113
IV-9	Cumulative Grade Point Averages of 1961-62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, 1961-1962	115
IV-10	Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American	116
IV-11	Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American	
IV-12	Cumulative Grade Point Averages of 1961-62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, 1961-1965	118
IV-13	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test Who Attended the 1961 Summer Session, The University of Mississippi	119
IV-14	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, The University of Mississippi,	120
IV-15	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, The University of Mississippi, End of Second Semester	121
IV-16	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test Who Attended the 1962 Summer Session, The University of Mississippi	122

TABLE		PAGE
IV-17	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, The University of Mississippi, End of Third Semester	.123
IV-18	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, The University of Mississippi, End of Fourth Semester	.124
IV-19	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test Who Attended the 1963 Summer Session, The University of Mississippi	-125
IV-20	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, The University of Mississippi, End of Fifth Semester	.126
IV-21	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, The University of Mississippi, End of Sixth Semester	-127
IV-22	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test Who Attended the 1964 Summer Session, The University of Mississippi	-128
IV-23	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, The University of Mississippi, End of Seventh Semester	129
IV-24	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, The University of Mississippi, End of Eighth Semester	130

TABLE		PAGE
IV-25	Grade Point Averages and Disposition of 1961- 62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test Who Attended the 1965 Summer Session, The University of Mississippi	. 131
IV-26		-
IV-27	62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, The University of Mississippi,	. 135
IV-28	62 Entering Freshmen with Composite Standard Scores of Greater Than 18 on the American College Test, by Test Scores and by End of	136
V-1	Semesters of Attendance at the University of Mississippi for 1961-62 Entering Freshmen Based on Composite Standard Scores of the American College Test	147
V-2	Graduation and Disposition of 1961-62 Entering Freshmen Based on Composite Standard Scores on the American College Test, The University of Mississippi	
V-3-A	Mean Scores, Standard Deviations, and Correlations for Grade Point Averages of 1961-62 Entering Freshmen Based on Composite Standard Scores of the American College Test, for 1961-62, The University of Mississippi	150
V-3-B	Mean Scores, Standard Deviations, and Correlations for Grade Point Averages of 1961-62 Entering Freshmen Based on Composite Standard Scores of the American College Test, for 1962-63, The University of Mississippi	151
V-3-C	Mean Scores, Standard Deviations, and Correlations for Grade Point Averages of 1961-62 Entering Freshmen Based on Composite Standard Scores of the American College Test, for 1963-64, The University of Mississippi	152

TABLE		PAGE
V-3-D	Mean Scores, Standard Deviations, and Correlations for Grade Point Averages of 1961-62 Entering Freshmen Based on Composite Standard Scores of the American College Test, for 1964-65, The University of Mississippi	
VI-1	Disposition of the 1961-62 Entering Freshmen by Sex, on August 22, 1965, The University of Mississippi	. 158

Chapter I

INTRODUCTION

The increasing enrollments in colleges and universities across the country have caused college and university administrators and their boards to become concerned. Student selection has become a necessity for most schools with high school grades and test scores serving as the principle screening devices. More recently standardized test results have been used with greater authority to determine who should be admitted to universities of higher learning.

Beginning in 1961-62 the Mississippi Board of Trustees of Institutions of Higher Learning required that the American College Test be taken as a prerequisite for admission to state institutions. The Board permits each institution under its jurisdiction to set a minimum score on the test as a basis for denying admission to freshmen to the institution. The composite standard score of 15 was set by The University of Mississippi as the lowest score permitted for the admission of students in the 1962-63 school year from Mississippi students. The minimum admission score set for out-of-state students was 19.

I. THE PROBLEM

Since the minimum score was not applied for students seeking admission in 1961-62, it was thought that a study to determine the success of the entering freshmen for 1961-62 based on the scores earned on the American College Test



during that year would be of benefit to the University in studying the relationship of student scores on the ACT and student progress at the University.

II. THE ACT TEST

From the 1964-65 Student Information Bulletin the American College Test has been described as follows:

The American College Test (ACT) Program is an admissions, scholarship, guidance, and placement test battery designed to provide helpful information and service to colleges, to high schools, and to you, the student, at a time when it will be most useful.

The ACT battery has two parts. The major part consists of four tests-in English, mathematics, social studies, and the natural sciences. The second part is the Student Profile, which is not a test at all, but rather a series of questions about your extracurricular activities and educational plans.

The purposes of the four tests are explained by saying:

The English examination is primarily a test of appropriateness and effectiveness of written expression. Most items are concerned with elements of diction, style, phraseology, form, and organization; other items are related to the correctness and mechanics of writing. The mathematics test is concerned with many of the mathematical skills and concepts taught in high schools.

The tests in social studies and natural sciences are designed to measure the ability to perform the kind of reasoning and problem solving required in these fields. They also include items that test knowledge of information sources in these fields and capacity for special study.

A total score on these four test provides an estimate of your ability to succeed academically in college. 2

lThe American College Testing Program, 1964-65 Student Information Bulletin (Iowa City, Iowa: The American College Testing Program, Inc., 1962) p. 2.

²Ibid.

Further explanation was given concerning the use of the test scores by the following statement:

Your scores on the ACT tests are reported to your high school and yourself for use in future decisions about your college plans.

The scores are also reported to the colleges you designate and may be used in a number of different ways:

To help them decide whether you should be encouraged to attend.

 To help them decide whether you should be granted a scholarship, a student loan, or any special award or help.

To communicate with you to help make arrangements for admission and registration.

To help in deciding your placement in English and mathematics curricula.

 To help determine in which sections of freshman English and mathematics you should be enrolled.

To help advise you about your educational goals, the courses you should take, and how heavy a load to carry.

Not all colleges use this information for all of these purposes, but all colleges requesting the scores use them in at least one of these ways. These colleges are therefore anxious for you to take the tests if you are considering the possibility of attending.

III. ORGANIZATION OF THE STUDY

In this study the students of the 1961-62 entering freshman class have been separated into three groups based on the scores earned on the American College Test: These groups include: (1) students with a composite standard score of less than 15, (2) students with a composite standard score of 15, 16, 17, and 18, and (3) students with a composite standard score of more than 18.

ERIC

^{3&}lt;u>Ibid.</u>, p. 3.

Chapter I gives an analysis of the original class in regard to sex and residence; i.e., indicating students by sex from Mississippi and from out-of-state. The percentiles, by sex, on the basis of separate subject matter fields are indicated as well as percentiles for the total composite standard scores.

Chapter II presents an analysis of the students with composite standard scores of less than 15 by showing the enrollments, by schools, by semesters, the grade point averages, the disposition of students, the mean grade point averages, and the standard deviation for each ACT score through the four year period.

Chapter III presents the same data for the students who were admitted with composite standard scores of 15, 16, 17, and 18. This division was made since the out-of-state students were required to make 19 for admission.

Chapter IV presents the same data for students included among the 1961-62 entering freshmen who had composite standard scores of better than 18.

Chapter V gives an analysis of the retention of the students of the entering class of 1961-62 based on the composite standard scores of the American College Test. The data show the number of semesters that students remained at the University, the number of students graduating, and the time of graduation.

Chapter VI presents a summary of the report and conclusions concerning the class.



IV. ANALYSIS OF THE 1961-62 ENTERING FRESHMEN

In the 1961-62 entering freshman class there were 1062 students for which scores on the American College Test (ACT) were available for all but nine. For these nine, scores were available on the College Qualifying Test (CQT). The scores of the College Qualifying Test were transformed to equivalent ACT scores by use of the percentiles given for each. However, by observation of the scores made by a number of students who had scores on both, it was found that scores on the CQT test were slightly higher than scores on the ACT test. For this reason the percentiles presented in this chapter have been determined on 1053 students who had scores on the ACT test. This accounts for the variance in number of individuals who have been counted for making the percentiles and for those who are included in school enrollments and grade reports which are included in subsequent chapters.

Table I-1 presents the freshman enrollments by sex and residence for 1961-62 entering freshmen based on American College Test composite standard scores.

Tables I-2 - I-5 present the percentiles in the four subject areas of English, mathematics, social studies, and science and Table I-6 presents the percentiles for composite scores for the group under study. The figures vary slightly from those issued from the Division of Student Counseling of The University of Mississippi because this study includes only the freshmen scores but the percentiles given by the Counseling Center included a number of upperclassmen who took the ACT for admission.

TABLE I-1

FRESHMEN ENROLLMENTS BY SEX AND RESIDENCE OF 1961-62 ENTERING FRESHMEN BASED ON THE AMERICAN COLLEGE TEST COMPOSITE STANDARD SCORES

ACT Score	M	lasia: Resid	sippi ents		t-of-3 eside:		P.,	Tota	1
	. M	·F	ज्या । छ	M	E aprolei	T	M	rollm	ENTS
(1)	(2		$(\overline{4})$	(5)	(6)	(7)	(8)	(9)	(10)
Below					•				
10	٥		. 9		9		0		7.7
ĩŏ	9 5 9 8 6	9	7		2 1	2	. 9	2	11
· iĭ		2 4 5 6	13	•	Т	į	5	3	8
.19	9	2		2 6	3 1	. 5 7	11	7	18
12 13) 6.	13	D	Ţ	7	14	6	20
14	12		12	3	- 2 7	. 5 13	. 9	8	17
15		4	16	6	7	13	18	11	29
16	21	12	33	10	8	18	31	. 20	51
	24	12	36	1.4	13	27	38	25	63
17	26	13	39	10	14	24	36	27	63
18	29	19	48	11	14	25	40	33	73
19	35	26	61	. 24	23	47	59	49	108
20	33	19.	52	20	12	32	53	31	84
21	28	17	45	22	15	37	5 0	32	82
22	44	15	59	23	. 17	4 0	67	32	99
23	.24	21	45	17	13	30	41	34	7 5
24	27	14	41	16	7	23	43	21	64
25	30	9	39	12	5	17	42	14	56
26	19	10	29	11	- 5	16	30	15	45
27	18	- 6	24	5	8	13	23	14	37
28	15	2	17	4	1	5		3	22
29	8.	3	11	4 1	1	2	_9	4	13
30	10	1	11		5 8 1 1	2 1	1Ŏ	4 2	12
31	8	•	8			. –	-8		8
32	4		4				4		4
Cotals	452	220	672	217	173	390	669	393	1062

ERIC

TABLE 1-2

PERCENTILES FOR ACT ENGLISH SCORES
OF 1961-62 ENTERING FRESHMEN
THE UNIVERSITY OF MISSISSIPPI

		ales	Fe	males	1	otal
Score	N	7-tile	A.	7-tile	N	7-tile
<u>(1)</u>	(2)	(3)	(4)	(5)	(6)	(7)
36	1	99.9			1	99.9
32	•••	00.0	9	99.7		99.8
31 .	3	· 99. 6	2 2 5	99	2 5 8	99.4
30	3	99	5	98	ĕ	99
29	12	98	15	96	27	97
28	12 13	96	16	92	29	95
27	8	95	25	87	33	92
26 25	30	92 88	3 5	79	65	92 87
2 5	25	88	39	7 0	64	81
24	36	83	24	61	6 0	75
23	37	77	56	51	93	68 ·
22	67	7 0	28	41	95	59
21	55	6 0	35	33	9 0	5 0
2 0	48	52	28 35 31	2 4 16	79	42
20 19 18	57	45	31	16	88	34
18	63	35	11	11	74	2 6
17	42	52 45 35 28 22 18	31 11 15 9 5 2 2	8	57	20
16	33	22	9	4	42	15
15 14	26	18	5	3	31	12
13	19 17	14	2	2	21	9
12	7	11	Z	4 3 2 1 1	19	12 9 8 6 4
11	, E	9 8 7 5	Z	1	9	6
10	6 17	7			6 17	6
-9	11	,			11	•
•	8	•			11	3
8 7 6 5	4	3 2 2			0	2 2
6	Ä	2			Ā	2
5	Š	ĩ	1	.3	Ā	†
4	4 3 2	ī	-		2	Ť
. 3	_	_			4	•
. 3 2 1	1	1			1	. 4
1	3	• 2	1	.1	4	:ī
Totals	661		392	•	1053*	*

^{*}Nine students took the CQT but not the ACT and are not included in this table.

TABLE I-3

PERCENTILES FOR ACT MATHEMATICS SCORES
OF 1961-62 ENTERING FRESHMEN
THE UNIVERSITY OF MISSISSIPPI

	Males			males	Total		
Score	N	7-tile	N	7-tile	N	7-tile	
77/	(2)	(3)	(4)	(5)	(6)	(7)	
3 6	1	99. 9			1	99. 9	
35	-	99	1	99.8	7	99.5	
34	6	98	_	0010	6	99	
33	10	97	1	99.6	11	98	
32	15	95	1	99	16	97	
31	9	94	1 1 2 2 7	99	11	96	
30	15	92	2	98	17	94	
29	23	89	7	97	30	92	
28	22	85	10	95	32	89	
27	35	81 76	8	93	43	.86 82	
2 6 2 5	3 0 33	76 71	10 10	91 88	4 0 43	78	
24	3 1	67	12	85	4 3	76 74	
23	34	6 2	13	. 82	47	69	
22	34	57	19	78	53	65	
21	52	50	18	73	70	59	
20	42	43	24	68	66	51	
19	4 8	3 6	37	60	85	4 5	
18	5 5	28	38	51	93	37	
17	44	21	29	42	73	29	
16	18	16	25	3.5	43	23	
15	30	13	25	29	55 45	19	
14	24	8 6	21	23 17	4 5 3 8	14	
13 12	15 8	8 6 4	23 15	12	23	10 7	
		-		า้ก็	7	<u>-</u>	
10	2 5	9	3	_9 _9	-	5	
_ 9	ĭ	3 2 2 2	12	10 9 7	8 13	4	
8	3 3	2	9	_	12	3	
7		_	1	3		2	
6	3	1	5	2	8	2	
5	3 1 3	<u>1</u>	5 3 12 9 1 5 1	4 3 2 1 1	1 8 2 6	6 5 4 3 2 2 1	
4	3	1	3	T	б	1	
3	•	•	•	9	E .	•	
11 10 9 8 7 6 5 4 3	3	. 2	2	.3	5	• 2	
±							
Totals	661		392		1053*	.•	

^{*}Nine students took the CQT but not the ACT and are not included in this table.

TABLE 1-4

PERCENTILES FOR ACT SOCIAL STUDIES SCORES
OF 1961-62 ENTERING FRESHMEN
THE UNIVERSITY OF MISSISSIPPI

ACT.		ales		males	T	otal
Score	(3)	7-tile	N	%-tile	N	%-tile
36 35 31 30 28 27 26 24 22 21 20 19 18 17 16 16 16 16 17 16 16 17 16 17 16 17 16 17 16 17 16 17 16 17 16 17 17 18 17 18 18 18 18 18 18 18 18 18 18 18 18 18	1 27 19 22 30 18 39 45 28 27 55 28 54 29 33 31 38 20 11 3 12 1	7-tile (3) 99.9 99.6 99.9 97.94 90.86 82.76 70.66 60.54 47.40 34.30 25.20 15.12 86.4 3.21	1 28 6 17 12 24 22 16 19 27 23 22 24 20 27 206 4 5 5 1 1 2 2 1	7-tile (5) 99.8 99.4 98.9 96.93 90.85 79.74 70.64 58.52 45.38 31.26 22.16 10.6 5.4 2.2 11.1	1 39 27 28 47 30 63 64 46 22 57 74 58 46 19 11 12 12 4 23 22 22	7-tile (7) 99.9 99.7 99.7 99.7 99.7 99.8 88.8 77.7 72.6 61.5 5.4 9.4 2.3 6.1 5.5 4.9 4.2 3.6 3.0 2.5 2.1 1.5 1.0 7.5 4.3 2.1 1.1 1.3 2.1 1.3 2.1 1.3 2.1 1.3 2.1 1.3 2.1 1.3 2.1 3.1 3.1 3.1 3.1 3.1 3.1 3.1 3.1 3.1 3
ĩ	1	.1			1	.1
Totals	661		392	,	1053*	·

^{*}Nine students took the CQT but not the ACT and are not included in this table.

TABLE I-5

PERCENTILES FOR ACT NATURAL SCIENCE SCORES
OF 1961-62 ENTERING FRESHMEN
THE UNIVERSITY OF MISSISSIPPI

ACT		ales	Fe	males	<u> </u>	otal
	(3)		N		N	%-tile
Score (1) 36 354 332 310 298 276 54 222 221 209 187 165 143 211 10 98 7 65 4 3 2 1		99.9 99.4 98.9 98.9 98.9 98.9 98.9 98.9		Males 7-tile (5) 99.7 99.3 99.3 99.98 96.99.99 86.81 75.68 62.55 48.13 10.8 6.5 3.2 1.8 1.3 1.8 6.5 3.2 1.8 1.8 1.8 1.8 1.8 1.8 1.8 1.8 1.8 1.8		
3 2 1	1	.1	1	.1	1	.1
Totals	661		392		1053*	

^{*}Nine students took the CQT but not the ACT and are not included in this table.



TABLE I-6

PERCENTILES FOR ACT COMPOSITE SCORES

OF 1961-62 ENTERING FRESHMEN

THE UNIVERSITY OF MISSISSIPPI

ACT		ales	Fe	males	<u> </u>	otal
Score	N (2)	%-tile (3)	<u>N</u> (4)	%-tile (5)	<u>N</u> (6)	%-tile (7)
36 35 35 32 30 28 27 26 27 26 27 26 27 20 18 17 16 15 11 10 19 87 65 43 21	480999999999999999999999999999999999999	99.6 99.8 98.96 94.91 87.81 75.68 60.51 44.36 28.23 17.12 97.5 3.21 1.3 .07	2 4 3 14 15 17 30 32 50 32 32 27 26 20 10 9 6 7 3	99.7 99.98 96.92 87.85 78.69 61.51 40.32 25.18 12.86 4.21	4 8 12 13 22 37 44 56 47 59 99 90 72 61 51 21 21 21	99.8 99.98 97.95 93.89 84.79 72.63 55.46 37.30 23.18 12.86 4.31 11.1.4.3.1.04
Totals	661		392		1053*	•

^{*}Nine students took the CQT but not the ACT and are not included in this table.



Chapter II

ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF LESS THAN 15

When the American College Test was first administered at The University of Mississippi to entering students in 1961-62 there was no minimum admission score set for entering freshmen. For this year, there were 103 students with scores of less than 15 who maticulated at The University of Mississippi and made up 9.7 per cent of the total enrollment of 1062 students. With the minimum score for admission set at 15 in 1962-63 this group would have been denied admission had they applied for entrance one year later.

Tables II-1 through II-8 indicate the enrollments by schools for all freshmen who were included in this group (below 15 on the ACT) for each semester. Summer session enrollments are not given because of the limited number of students from the group under study who were enrolled during this period. Changes in classification of students in the group caused by attendance in the summer sessions are noted in the data presented for the semester following the summer session.

Tables II-9 through II-12 show the cumulative grade point averages prepared for annual periods for preceding times of enrollment. In Table II-9 the summer sessions of 1961 and of 1962 are included in the first year's achievement. In Tables II-10 through TI-12 the reports are prepared at the end of the

summer session following an academic year and include all the grades earned prior to that time.

Tables II-13 through II-25 indicate the performance and disposition by semesters and summer sessions, by individual scores, of the 103 students who entered the 1961-62 freshman class with composite standard scores of less than 15.

Table II-26 indicates the changes in school emrollments for the group through the entire period of enrollment at The University of Mississippi.

Tables II-27 through II-28 summarize the grade point averages and disposition by semesters and summer sessions of the students based on composite standard scores of the American College Test.

TABLE II-1

INITIAL ENROLLMENT OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF FIRST SEMESTER

		College	om Sahoo'			
Standard Score	Liberal Arts	Bus. & Gov't.	or Schoo Educa- tion	l of Regist: Engi- neering	Phar- macy	Total
(1)	(2)	(3)	(4)	(5)	(6)	(7)
Below 10	3	4	3		1	11
10	1	3	4			8
11	5	6	6	1		18
12	9	5	5,	· 1		20
13	8	3	4	2		17
14	6	9	9	3	2	29
Totals	32	30	31	7	3	103
Total Freshmen Enrollments ir Schools	5 74	229	126	103	30	1062
Per cent in each school with scores less than 15	5.6	13.1	25.4	6 . 8	10.0	9.7

TABLE II-2

ENROLLMENT OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI, BEGINNING OF SECOND SEMESTER

Standard	Liberal	College Bus. &	Educa-	Engi-	Phar-	Total
Score (1)	Arts (2)	Gov't. (3)	tion (4)	neering (5)	macy (6)	(7)
Below 10	1	4	1		1	7
10		1	4			5
11	3	5	6	1		15
12	. 4	4	7	1		16
13	5	1	5	2		13
14	7	9	8		1	25
Totals	20	24	31	4	2	81

I. CLASSIFICATION BY SCHOOLS

At the beginning of the second year the group under study had dropped from 103 to 51, or to 49.5 per cent, of its initial enrollment, as is indicated in Table II-3. Of the 32 students enrolled in the College of Liberal Arts at the beginning of the first semester of 1961-62, 14 remained in the University at the opening of the third semester. Of these 14, two had transferred to the School of Business and Government, three to the School of Education, and nine remained in the College of Liberal Arts. One student transferred to the College of Liberal Arts from the School of Engineering. Thus 18, or 56.25 per cent, had dropped out of school from the College of Liberal Arts since the opening of the preceding year.

The School of Business and Government had an initial enrollment of 30 students. At the opening of the first semester of 1962-63, 11 students of the original group remained in the School of Business and Government, four had transferred from the School of Business and Government to the School of Education, two transferred from the College of Liberal Arts to the School of Business and Government. Of the initial group, 15, or 50 per cent, had dropped out of the University since the opening of the preceding year.

The School of Education retained 16 of its original 31 students in the group under study. There were three students who transferred from the College of Liberal Arts and four students who transferred from the School of Business and Government to the School of Education. The School of Education.

tion lost 15, or 48.4 per cent, of its initial 31 through drop outs.

The School of Engineering retained three of its initial seven students and one transferred to the College of Liberal Arts. This indicates the loss from the University for this group was three, or 43 per cent, of the initial group.

The School of Pharmacy retained two of the three students who enrolled in the school as freshmen in 1961-62.

One, or 33 1/3, per cent of the initial group was no longer enrolled at The University of Mississippi.

TABLE II-3

ENROLLMENT OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI, BEGINNING OF THIRD SEMESTER

Standard Score	Liberal Arts	Bus & Gov't	or School Educa- tion	of Registers Engi- neering	Phar- macy	Total
(1)	(2)	(3)	(4)	(5)	(6)	(7)
Below 10		1	3		. 1	5
. 10			3			3
11	2	4	3	1		10
12	3	2	4	1		10
13	2	2	4	1		9
14	3	4	6		1	14
Totals	10	13	23	3	2	51

ERIC

TABLE II-4

ENROLLMENT OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI, EEGINNING OF FOURTH SEMESTER

Standard	Liberal	College Bus. &	Educa-	ol of Reg: Engi-	istration Phar-	Total
Score (1)	Arts (2)	Gov't. (3)	tion (4)	neering (5)	macy (6)	(7)
Below 10		1	3		1	5
10			3			3
11	1	2	4	1		8
12	3	2	3			8
13	2	2	4			8
14	3	3	6		1	12
Totals	8	10	23	1	2	44

The state of the s

Table II-5 indicates that of the initial 103 students with composite standard scores of less than 15 on the American College Test who had enrolled two years previously, only 37, or 35.9 per cent, enrolled at the beginning of the 1963-64 school year. Although some students of the initial enrollment had changed schools in the University, there were remaining only three of the initial 32 in the College of Liberal Arts, eight of the initial 30 in the School of Business and Government, 12 of the initial 31 in the School of Education, none of the original seven in the School of Engineering, and one of the initial three in the School of Pharmacy.

The tendency of the group of students who entered The University of Mississippi as freshmen in 1961-62 with composite standard scores of less than 15 on the American College Test, was to move from the College of Liberal Arts to the professional schools. Seven of the students in the School of Education had originally been in the College of Liberal Arts. There were two transfers to the School of Education from the School of Business and Government, and one transfer each from the Schools of Education, Engineering, and Pharmacy to the School of Business and Government.

TABLE II-5

ENROLLMENT OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI, BEGINNING OF FIFTH SEMESTER

Standard Score	Liberal Arts	College Bus. & Gov't.	or School Educa- tion	of Regis Engi- neering	tration Phar- macy	Total
(1)	(2)	(3)	(4)	(5)	(6)	(7)
Below 10		1	2		1	4
10		•	2			2
11		3	4			7
12	1	1	4	٠		6
13		1	3			4
14	2	5	7			14
Totals	3	11 .	22		1	37

TABLE II-6

ENROLLMENT OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI, BEGINNING OF SIXTH SEMESTER

Standard Score	Liberal Arts	College Bus. & Gov't.	or School Educa- tion	of Register Bright - neering	Phar- macy	Total
	(2)	(3)	(4)	(5)	(6)	(7)
Below 10		1	1		1	3
10			1			1
11		2	3			5
12	1	1	4			6
13		1	3			4
14	2	4	7			13
Totals	3	9	19		1	32

At the beginning of the fourth year the group of students under study had dropped from an initial enrollment of 103 students to an enrollment of 33, or 32.0 per cent, of its initial enrollment. During the third academic year of enrollment the loss was from 37, who were enrolled at the beginning of the fifth semester, to 33 for the beginning of the seventh semester, a net loss of 4 students. However, the actual loss from the beginning of the fifth semester to the beginning of the seventh semester was nine students. Five students who had previously been enrolled at the University but were not enrolled at the beginning of the fifth semester returned at the beginning of the seventh semester.

Of the initial 32 students in the College of Liberal Arts, there were six remaining at the beginning of the seventh semester. This represents an increase of three over the preceding year's enrollment caused by the return to the University of four students who originally were enrolled, but not enrolled at the beginning of the fifth semester in the College of Liberal Arts, and the withdrawal of one student from the college after the beginning of the fifth semester.

Of the initial 30 students in the School of Business and Government, only five remained at the beginning of the seventh semester. There were transfers of one student from each of the Schools of Education, Engineering, and Pharmacy to the School of Business and Government. There were no transfers after the beginning of the fifth semester although the school lost three students by withdrawal from the University.

In the School of Education there was an initial enrollment of 31 students of which only ten remained at the beginning
of the seventh semester. Eight students had transferred from
other Schools since initial enrollment, as follows: College
of Liberal Arts, five; School of Business and Government, two;
and School of Engineering, one via College of Liberal Arts.
There were no changes in schools either to or from the School
of Education. There were five withdrawals and one re-entry
after the beginning of the fifth semester and at the beginning
of the seventh semester.

In the original enrollment of seven students in the School of Engineering, not a single student remained at the beginning of the seventh semester although one student was still enrolled at The University of Mississippi in the School of Business and Government.

There was an initial enrollment of three in the School of Pharmacy. Only one remained in the School of Pharmacy at the beginning of the seventh semester. One student transferred to the School of Business and Government and one withdrew.

TABLE II-7

ENROLLMENT OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI, BEGINNING OF SEVENTH SEMESTER

Standard Score	Liberal Arts	College Bus. & Gov't.	or School Educa- tion	of Regist Engi- neering	Phar- macy	Total
(1)	(2)	(3)	(4)	(5)	(6)	(7)
Below 10		1	2		1	4
10			1			1
11		3	3			6
12	···. 2	1	3			6
13	ĺ	1	3			5
14	3	2	6,			11
Totals	6	8	18		1	33

TABLE II-8

ENROLLMENT OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI, BEGINNING OF EIGHTH SEMESTER

Standard Score	Liberal	College Bus. & Gov't.	or School Educa- tion	of Registers Engi- neering	Phar- macy	Total
(1)	Arts (2)	(8)	(4)	(5)	(6)	(7)
Below 10	1	1			1	3
10						· .:
11		2	2			4
12	1	1	2			4
13	1.	1	2			4
14	. 3	2	6			11
Totals	6	7	12		1	26

II. STUDENT ACHIEVEMENT

When the students under study entered the 1961-62 freshman class at The University of Mississippi the grade point average was based on a five point grading system. All grades which are used in this study have been translated to the four point grading system which was initiated at the beginning of the fifth semester following initial enrollment of these students, or the first semester of 1963-64.

Tables II-9 through II-12 show the annual cumulative grade point averages, the mean grade point average, the standard deviation for individual composite standard scores of less than 15 on the American College Test, and the correlation coefficient between the cumulative grade point averages and the ACT scores. Tables II-13 through II-25 indicate like data for each semester and summer session and the disposition of the students during each enrollment period.

⁴The formula which was used to translate the five point grade-point-average to the four point grade-point-average was as follows: New Grade Points = Total Grade Points - (No. Hours Attempted + No. Hours Failed).

28

TABLE II-9

CUMULATIVE GRADE POINT AVERAGES OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST, 1961-1962 THE UNIVERSITY OF MISSISSIPPI

Standard Score	No.	Below 1.00	1.00- 1.49	1.50- 1.99	Grade Point - 2.00- 2.49	2.50- 2.99	3.00- 3.49	3.50- 4.00	Mean	g.D.
· (I)	(2)	(3)	(4)	(2)	(9)	(4)	(8)	(6)	(10)	
Below 10	10	2	က	Н	Н				1.01	.67
10	. &	က	2	П	Н	Н			1.32	. 94
11	18	7	S	S		ч			1.20	.55
12	20	œ	7	4			٦		1.03	. 82
13	16	S	က	4	က	1		r	1.43	.74
14	27	S	∞	ဃ	83	4			1.46	.72
Totals	100	33	28	23	7	7	1			**

TABLE II-10

CUMULATIVE GRADE POINT AVERAGES OF 1961-62 ENTERING FRESHMEN
WITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE
AMERICAN COLLEGE TEST, 1961-1963
THE UNIVERSITY OF MÍSSISSIPPI

				Gre	Grade Point Average	Average				
Standard Score	No.	Below 1.00	1.00- 1.49	1.50- 1.99	2.00- 2.49	2.50- 2.99	3.49 3.49	3.50- 4.00	Mean GPA	S.D.
(I)	(2)	(3)	(₹)	(5)	(9)	(4)	(8)	(6)	(10)	(EE
Below 10	5		7	Н	2				1.76	68.
10	က			H	н	7			2.20	.34
11	10	Н	လ	က		٦			1.52	.47
12	10	· ~	ග	87			1	. •	1.57	.61
13	6	က ·	н	က	-	Н			1.53	69.
14	15	Н	4	4	လ	ч			1.73	. 58
Totals	53	9	18	14	6	4	1			

TABLE II-11

CUMULATIVE GRADE POINT AVERAGES OF 1961-62 ENTERING FRESHMEN
WITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE
AMERICAN COLLEGE TEST, 1961-1964
THE UNIVERSITY OF MISSISSIPPI

				Gre	Grade Point	- 15				
Standard Score	No.	Below 1.00	1.49	1.50- 1.99	2.00- 2.49	2.50- 2.99	3.49 3.49	3.50- 4.00	Mean GPA	S.D.
(1)	(2)	(3)	(4)	(5)	(9)	(3)	(8)	(6)	(10)	(11)
Below 10	ന		٦		87				1.95	.47
10	8	٠		1	H				2.04	88
11	9	٦	87	භ					1.46	en. ♣1
12	7	8	က	୍. ଦୀ					1.44	. 24
13	4			က	Н				1.92	.18
14	13		₹'	လ	4				1.78	36
Totals	35	ബ ബ	10	14	ω				·	

TABLE II-12

CUMULATIVE GRADE POINT AVERAGES OF 1961-62 ENTERING FRESHMEN
WITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE
AMERICAN COLLEGE TEST, 1961-1965
THE UNIVERSITY OF MISSISSIPPI

				Gre	Grade Point	E Average	a)			
Standard Score	No.	Below 1.00	1.00-	1.50- 1.99	2.00- 2.49	2.50- 2.99	3.49	3.50- 4.00	Mean GPA	S.D.
~ (I)	(2)	(3)	(4)	(5)	(9)	(4)	(8)	(6)	(10)	
Below 10	4	Н	М		83				1.59	.71
10	7				,				2.41	•
11	9	П	2	က					1.43	. 32
12	9		8	4					1.61	.21
13	د			က	7				1.96	.13
14	12	Н	7	က	9	7			1.87	.48
Totals	34	က	9	13	11	П				

TABLE II-13

GRADE POINT AVERAGES OF 1961-62 ENTERING FRESHMEN
WITH COMPOSITE STANDARD SCORES OF LESS THAN
15 ON THE AMERICAN COLLEGE TEST WHO
ATTENDED THE 1961 SUMMER SESSION
THE UNIVERSITY OF MISSISSIPPI

* 77.				Grad	e Point	Average				
Standard Score	No.	Below 1.00	1.00-	1.50- 1.99	2.00- 2.49	2.50- 2.99	3.49	3.50- 4.00	Mean	S.D.
(1)	(2)	(3)	(4)	(2)	(9)	(4)	(8)	(6)	(10)	
Below 10	Н				1				2.00	(See
10										(alou
11	Н		Н						1.00	
12	က	ı	<u>.</u>	-					1.08	
13	භ	82			1				.97	
14	4			63	٠.	,н	Н		2.19	,
Totals	12	က	2	ဧာ	2	1	. 1			

of students enrolled during the summer of 1961, standard deviations and the correlation have not been determined. number Due to the small the standard devi Ö

Not

e «

TABLE :-14

ERIC Full Text Provided by ERIC

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 '
ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST,
THE UNIVERSITY OF MISSISSIPPI
END OF FIRST SEMESTER

Stan-							63	Grade P	Point Average	rerage			
	No.	With- drawals	Dis- missals	Proba- tions	Below 1.00	1.00- 1.49	1.50- 1.99	2.00- 2.49	2.50- 2.99	3.49	3.50- 4.00	Mean*	S.D.
	(2)	(3)	(4)	(5)	(9)	(3)	(8)	(6)	(10)	Œ	(12)	(13)	114
	,	•	4	ı	C	•	•	•				Ċ	(
	Ţ	i	n	ഹ	_		-	1				, 88	0
	ω		83	87	4	Ч	7				~	1.29	1.12
	18		च्या	, φ	9	œ	က	Н				1.08	.58
	20		9	œ	11	2	ಉ			Н		.98	.74
	1.7	7	Ø	လ	လ	က	ෆ	က	87			1.44	. 84
	53	8	က	φ	9	7	7	ෆ	4			1.47	.75
	Totals103	4	21	34	39	25	18	6	9	-	-		
1	-												

*N = Col. 2 - Col. 3.

TABLE II-15

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62

RNTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST,
THE UNIVERSITY OF MISSISSIPPI
END OF SECOND SEMESTER

	S.D.	(14)	88	.61	. 56	.84	.61	.72	
	Mean	(13)	1.46	1.78	1.37	1.30	1.70	1.58	
	3.50- 4.00	(12)			•	-			1
Average		(11)				n			
		(01)	H	-	Н		1	8	9
Grade Point	2.00- 2.49	6	H	8	т	Н	4	ø	15
9	1.50- 1.99	(8)	8		Ŋ	၈	Н	4	15
	1.00-1	(2)	بسب	~ 3	4	9	S	9	24
	Below 1.00	(9)	83		4	2	-	4	16
	Proba-	(5)			1		T	8	4
	Dis- missals	(4)	87	7	က	7	၈	Ŋ	24
	With-	(3)					Н	ഩ	4
	Ž	(2)	7	· M	15	16	13	. 22	Totals:81*
010	dard Score No.		Below 10	.00	11	12	13	14	Total

dismissal during the preceding semester. This readmission was secured by appeal. One student returned to The University of Mississippi after withdrawing during the preceding semester. enrollment of 81 instead of 78 might be expected from withdrawals and dismissals in Table II-14 is caused by readmission on probation of four students who received instead of 78 might be expected from withdrawals and dismissals ല indicated #11

34

TABLE II-16

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62
ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST
WHO ATTENDED THE 1962 SUMMER SESSION
THE UNIVERSITY OF MISSISSIPPI

	•		•			- 1	9	ij		K		
N	Wit	Dis- missals	Proba- tions	ቖӚ	1.90- 1.49	1.99	2.49	2 50 2 99	3.49	4 00 4	Mean	S.D.
2	(3)	(4)	(5)	(6)	(3)	(8)	(6)	(OT)	(11)	(12)	(13)	(14)
G			•				-	-			36 6	6
N				**			4	4			9	•
က	гH	 1		H			ط				1.18	68
Φ.			Н	8	Н	·	(53)	83	•	•	1.72	.87
ന				٠	H	H	,				1.72	.52
က		· H		-	. 1	~					1.20	•40
4	н	•			8	-		•		•	1.25	. 35
Totals 23	2	03	-	4	5	က	9	က				

TABLE II-17

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62
ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST,
THE UNIVERSITY OF MISSISSIPPI
END OF THIRD SEMESITER

Q+an-							Ö	Grade Point		Average			
dard	Q N	With-	Dis- missals	Proba-	Below	1.00-	1.50-	2.00- 2.49	(C)	3.49	3.50- 4.00	Mean	S.D.
	(2)	(3)	(4)	(5)	(9)	(3)	(8)	(6)	(10)	(11)	(12)	(13)	(14)
Relow	٠												-
10	S		•	,		83		H,	₩.			1.87	.61
10	က						•	ಉ		•		2.22	.03
Ħ	10	rel		က	ល	လ	·H			Н	: .	1.41	•64
12	10	83		က	က	Н	Н	87	•	·H		1.61	.71
8	ຫຼ	H	*	~	ಕು	Н	83	Н	Н			1.46	984
14	14		٦	· 4 1	က	က	N	<u>ਚ</u>	.~			1.58	. 80
Totals 51	51	4	5	12	11	12	9	11	5	7			

student was given disciplinary dismissal for one semester. *One

TABLE II-18

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62
ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST,
THE UNIVERSITY OF MISSISSIPPI
END OF FOURTH SEMESTER

With	Dis- Proba- Below I.	-00	1.50-	1 1.		Average 3.00-	3.50-		
ls missals	-	(7)	1.99	2.49	2.99	3.49	4.00	Mean (13)	S.D.
		83	٦	Н	Н			1.97	.56
	1	-		83		-		2.00	.62
. ד	3 .		က	н	~			1.58	.77
	-	Н	~	က				1.72	.47
. 2	က		83	Н	H	H		1.68	.87
8	H	4	Н	က	2	н		1.89	.68
2 3 6	7	ω	6	11	5	2			

TABLE II-19

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62
ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST
WHO ATTENDED THE 1963 SUMMER SESSION
THE UNIVERSITY OF MISSISSIPPI

3.D.			.51	.27	.67	.92	
Mean (13)			2.24	2.02	1.88	2.70	
3.50- 4.00 (12)						1	1
Average 3.00-3.49						Н	П
Point A 2.50-2.99 (10)			ri,		Н		2
Grade Pe - 2.00- 2.49 (9)			Н	 1	Н	гH	4
G1 1.50- 1.99 (8)			-	Н	Н	ط'	4
1.00-1.49					Н		н
Below 1.00 (6)	; ;			. •			
Proba- tions (5)						8	2
Dis- missals (4)		•					
With- drawals (3)							
No.			က	87	4	4	13
Stan- de.rd Score (1)	Below 10	TO.	11	12	13	14	Totals 13

TABLE II-20

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62
ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST,
THE UNIVERSITY OF MISSISSIPPI
END OF FIFTH SEMESTER

						Grac		1 1	Average			
With- Dis- Proba- No. drawals missals tions	als	Proba tions	ı	Below 1 1 00 1	00 - 49	1.50- 1.99	2.00- 2.49	2.50- 2.99	3.49	3.50- 4.00	Mean	S.D.
(3) (4)	1 1	(2)		(9)	5	(8)	(6)	(10)	(11)	(12)	(13)	(14)
4 2 1	÷	Ä			-		H			•	1.69	. 48
2 . 1							Н				2.38	
7 2 1	2 . 1	H		7	2	, 2 2	٦	Н			1.71	. 55
9	83	7			· —	٠ د					1.66	.19
4							7	8			2.43	• 08
14 1 1 3	1 3	၈			မ	9	2	2		·	1.78	.43
Totals 37 4 3 7	3 7	7		1	7	13	7	5				

TABLE II-21

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62
ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST,
THE UNIVERSITY OF MISSISSIPPI
END OF SIXTH SEMESTER

	S.D.	. 64		.52	.52	• 20	.61	
	Mean (13)	1.75	2.32	1.56	1.49	2.42	2.19	
	3.50- 4.00 (12)							
verage	3.00- 3.49 (11)						ရာ	က
oint A	2.50- 2.99 (10)	٦				7		22
Grade Point Average	2.00- 2.49 (9)		Н	8	8	က	S	13
6	1.50- 1.99 (8)	Н			Н		က	5
	1.00- 1.49 (7)	1		ന	8		67	8
	Below 1.00 (6)				٦			1
	Proba- tions (5)			87	87		Н	ro
	Dis- missals (4)	ч		Н	87		1	5
	With- drawals (3)						•	٠.
	No. (2)	က	⊣	လ	9	4	13	32
Stan-	dard Score	Below 10	10	11	12	13	14	Totals 32

TABLE II-22

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62
ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST
WHO ATTENDED THE 1964 SUMMER SESSION
THE UNIVERSITY OF MISSISSIPPI

						1 1 1							
		With-	Dis-	Proba-		1.00	1.50-	90 90	4100	H .	3.50-		
score (1)	NO.	drawals (3)	mıssais (4)	110ns (5)	(9) (9)	1.49	1 3 9 9 (8)	2.49 (9)	(10)	(11)	(12)	(13)	(1.4)
Below				• •	•	•			•				
0	6 2			*		•		-	r-i	•		2.25	. 25°
A	•							•	• · · · · · · · · · · · · · · · · · · ·	*	•		No.
	က			•		H	H	Ä	•			1.68	
<u>~</u>	ഹ		H	• • •			1	: : H	•	H		. eg	្រូវ
7					•	•	••	H		•		2.15	
	N	\	**	•		***	એ	• •	r l			1,94	, 44.°
Totals 14	4	.	4			H	4	73	. 2	-			

*One student was given a disciplinary dismissal

TABLE II-23

ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST,
THE UNIVERSITY OF MISSISSIPPI
END OF SEVENTH SEMESTER

S.D.	77-7	89.		48	04	9	80	
Mean	(OT)	1.94	2.80	1.66	30.2	96 T	2.10	
3,50- 4,00	126)	4 4 3,1						•
Average 3.00-	(77)					6.	7	, ₁ -4
Point Ave 0- 2.50- 3 9 2.99	(07)	.03	; H		H	-	41	6
2.0 2.4		•		.e.ɔ	က	က	ัณ	11.
1.50 1.99	0			H	-		က	S.
1.00		~ 1	1.					41
Below 1.00				H			ri e	ଜ୍ୟ
Proba- tions		•						
13	7#1	r-I		8	H			ဖ
With- drawals	9							
No.		4		G	ဖ	9	H	භ භ
Stan- dard Score	Bellow		9	7	, CV , CT	a .		Total.s

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62
ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST,
THE UNIVERSITY OF MISSISSIPPI
END OF EIGHTH SEMESTER

	S.D.			73	52	73	₹9. 7	
	Mean	(OT)	0	. T.	2.33	2.22	2.32	
. 1	4.00	227						
Verage	39 3 49					ન :		7
oint A	707				~ 3	H	Ó	-O.
Grade Point	2.49		.	Ä			2	4
کا اوران	000 000		,		rei			. ഔ
C C	1.49			₽,		—	1	က
e G	1.00			-1			-	6 2
	ropa-	S				rd		· က
	missals			7			, H	4
	drawals.							H
	No		2	4	4	4	4	. 26
Ties of		EET OW			2	27	*	Totals 26

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62
ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES
OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST
WHO ATTENDED THE 1965 SUMMER SESSION
THE UNIVERSITY OF MISSISSIPPI

	, d	(14)	. 12 90 90	
	Mean		1.2 50 25 1.3 25	-0.
*	1.	(12) (7 7 7	i
	(Y) 41	70		
Aver	3.00-3.00-3			•
Poin	2.50- 2.99		H	
Grade Point Average	2.00- 2.49	(6)	7 7	7
	1.50-	(8)		က
	1.00- 1.49			
	Below 1.00	(9)		H
	Proba-	(5		
	Dis- nissals	(4)	7	6)
	AB			
	With- drawals	(3)		
	No. 4	(3)	W 44, W 10	
Tank	dard Score			Totals

III. SUMMARY OF REPORTS

In the preceding tables of Chapter II, the data which indicate enrollments in schools for each semester and summer session for entering freshmen with composite standard scores of less than 15 on the American College Test have been presented. The enrollments indicate a drop for the group from an initial enrollment of 103 during the first semester of 1961-62 to 25 at the end of the second semester of 1964-65, or the end of eight semesters. Coefficients of correlation, though not an integral part of the study, range from highs of .295 and .259 during the sixth and eighth semesters to a low of -.311 during the 1962 summer session.

In the following section the tables which were presented in this chapter have been summarized.

Table II-26 presents summaries of enrollments by schools and by semesters.

Table II-27 shows a summary of the grade point averages and disposition, by semesters and summer sessions, of the 1961-62 entering freshmen with composite standard scores of less than 15 on the American College Test.

Table II-28 shows a complete summary of the grade point averages and disposition, by semesters and summer sessions, for individual scores of the 1961-62 entering freshmen with composite standard scores of less than 15 on the American College Test.

TABLE II-26

ENROLLMENT OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS AND BY SEMESTER, THE UNIVERSITY OF MISSISSIPPI 1961-62---1964-65

			Yea	r of I	Enrollme	ent			
	196	1-62		2-63		3-64	1.96	4-65	
School or	lst	2nd	15t	2nd	. 1st.	2nd	lst	2nd	
College		Sem.	Sem.		Sem.	Sem.	Sem.	Sem	
(1)	(2)	(3)	. (4).	(.5)	(6)	(7)	(8)	(9)	<u> </u>
Liberal Arts	32	20	. 10	8	3	3	6	6	
Business and Government	30	24	13	10	. 11	9	8	7	
Education	31	31	23	23	22	19	. 18	12	
Engineering	7	4	3	1					
Pharmacy	3	2	2	2	T.	.1	1	1	
	رماه هم المراقع المراق المراقع المراقع المراق				10 (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)				H. 187
Totals	103	81	. 51	44	37	32	33	26	
		in the factor of the second of	•		****	in the second			3.

Note: Data are for the beginning of semesters.

TABLE II-27

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN TITH COMPOSITE STANDARD SCORES OF LESS THAN 15 ON THE AMERICAN COLLEGE TEST, THE UNIVERSITY OF MISSISSIPPI 1961-62---1964-65

	1961	1.96	1961-62			962-63	က		P:		الساا	964-65	60
position Morres		Ist 2		l ki	1st Sem.	2nd Sem	S. S.	1st Sem		ကြီး	1st Sem.	Sem.	S.S.
		1	(4)	5)	(9)	(2)	(8#)	(6))			(12)	(23)	(14)
Snrollments	103		*18	6 7	75	4	F	16	6 7	17	***************************************	G.	87
# thdrawals		4	*	O.	ু বা	0		4				4	
Dismissals	. 01	21.2	24	~	***	က		, do	O	7	6		က
Probations		4	₹'	id H	27 77	ဖွ	6		S			03	
Grade Point Iverages:													
_ H 7			4	4 N	L2 12	~∞	—		μώι		અનું !	လေးဟ	
000	ന ന റ ്	ထတ္	ម្រា	ကယက	о П	는 이 니 다	440	& L K	က က က	4.N 64	റ പ്ര	গ বা তা	244
50-4 50-4		3 m m)		> 6 3) Ø	l pool pool		m			~	m
Graduates												4	6

^{*}Additional students enrolled who had not been during preceding semester.

^{**}One student received disciplinary dismissal.

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN
WITH COMPOSITE STANDARD SCORES OF LESS—THAN 15 ON THE AMERICAN
COLLECE TEST, BY TEST SCORES AND BY END OF SEMESTER,
THE UNIVERSITY OF MISSISSIPPI
1961-65

											4	8	16	
	י ט	(15)	(1,3)	Thus.		77	₩.	ุณ	, O				1	
	2nd::	(14)	(92)	က		4	4	4	-	(1)				
	lst Som		(33)	4	TH.	Ö	6	Ą	4					Estate of
	J v		(77)	64		6 0	. m		j.	3	\ . 5			
7 0 0	2nd Som		32) (က		io.		4	က်					
20 L	1st 2nd	الما.	(37) (4	O)		9.	4	4	(4)	Ņ			
	. بشر م مارید مارید و این													
63	מ		(13)			က်	07	4	41					
020	st 2nd		(44)	D	କ	&		ထ	지 대 :	(2)	e de general de la composition della composition			H
vi j	1st		(27)	വ	ęņ .	10	D.T	O,	14	(4)				M
	v.	(9)	(23,)	77	က	Φ.	က	က	4	(2)		H		
02 120	2nd 2nd	(2)	(81)	.	Ŋ	12	L O	ണ പ	2	(4)				
F	JE T	(4)	03	H	w	E	20	2	 00 20	(4)	H			
(30)	7 C	(3)	(12)(103)	H			က	က	4					
: 4														
	Enroll.	(2)	(103)	H	₩	M	20	17	တ် လ				1	20
10年 10年 10日		1 1	ats.						e de	als:				
	Disposition by Scores		Enrollments:	Below 10	O H		77	e H	7.	Mitherawals	ow 10	L L	d	7
100			E	Be] c						11.11	Below			

TABLE II-28 (continued)

1962-63 1963-64 1964-65 1st 2nd 1st 2nd Sem. Sem. S.S. Sem. Sem. S.S. Sem. Sem.	
2 deg. 5.5. Sem. S.5. Sem.	7 7 7
2nd 1st 2nd Sem. S.S. Sem.	7
2nd 1st Sem S.S. Sem (3) (2) (10) (2) (2) (2) (2) (11) (2) (2) (2) (2) (2) (3) (4) (5) (5) (5) (5) (6) (7) (6) (7) (7) (7) (7) (7) (7) (7) (7) (7) (7	7
2nd 2nd Sem. S.S. (8) (9). (6) (2)	
962-65 3	
解書、論論的 - 本書 - 書	
"我, 不以你只说: "(1961年)这次的连续第二人包含,我们用某个人就会的"我 想 ,但这个人"的人们的"我们"。	
	e T
361-62 Sem: S.S. (5) (6) (5) (6) 3 1 3 1 3 1 3 2 5 5 5 5 (4) (1)	Н
1st 2nd Sem. Sem. (4) (5) (2) (2) 2 3 3 3 2 3 2 4 5 6 7 6 7 3 3 3 5 3 5 3 5 3 5 3 5	,rl .co
1961 S. S. Sell 1 1 4 4 4 4 4 4 4 4	
	8
Disposition Englished Scores Medical 13 Linguissals: Lin	
GEA and Disposition 13 13 14 ELOW 10 12 12 12 12 14 *One studen Below 10 12 12 12 14 *One studen Below 10 10 10 10 10 10	1

	li H									Ţ.		
	.65 											
	1964-65 2nd Seπ S (14) (M
	1st Sen 113					•						
	(2) (2) (2)	i.										
	963-64 2nd Sem (11)	4				3						
	Sem (10)	Ŋ		69		Ĵ						
	. S (6)			67		Ver Maria	6 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -					
	962-63 2nd Sem (8)			7		6			c 7.	•	രാ	н.
ned)	lst Sem (7)	က		4					7	က	က	e
(contin	S S (6)					(4)			Ŋ			
11-28 (0	1961–62 2nd Sem 8 (5)			7	A	(1 <u>6</u>)	\ 0		₩.	မှာ	-1	4
	See (4)	ά	Ò	Φ.		(39)	_	1	-Ø		ល	ω
TA THE TANK	196 <u>1</u> \$ \$ (3)					(3) (39)			1		67	
	Initi	20.	ì	53				00	0	20	4	29
	i on				ŧ.	00						
	GPK and Disposition by Scores	37	2	A	Grade Point Averages:	Delow 1	Below 10	12	H		3	7
9_1/21			有			B	8					

ERIC Full Taxet Provided by ERIC

TABLE II-28 (continued).

			•	196			•.	*			• •		•	. 5		
	SS	(15)	*		· ·					(3)			; ;—l	 1		 1
964-65	2nd Sem.	(14)	(2)	[H			₩.	H	-4	(3)	H	•	•	H	H	•
19	, ,,	(13)	(4)	8	139 249 249	Н	r -1	•	•	(2)		4		H		ഇ
	S	12)	(1)	• .	· · •	H	•		,	(4)	• ,		~	·	*	62
3-64	2nd Sem. S) (77	(8)	·		က	87	s	87	(2)	Н		v .	H		က
196	ł •}	COT	(2)	· —		87	H·		က	(13)			8 7	3	:	ဖ
	S.S.	66)	(1)	•		•		П	•	(*)	•		· H	Н	Н	-
1962-63	2nd Sem		(8)	03	· H		H		4	(6)	Н		က	87	83	Н
	1st Sem.		(12)	87		ر م	ିଳା	, 1	က	(9)		٠,	7 -1	H	Ø	83
1	S.S.	(0)	(5)			 1.	, , ,	Н	62	(3)				H	н	H
961-62		(0)	(24)	1774		4	မ	2	9	(12)	62		വ	က	-	4
	u el	(4)	(22)		Н	œ	က	က	. 6	(18)	H	Н	က	ຕຸ	က	7
1961	S.S	(3)	(2)		•	m	ered Tered		· · · ·	(3)				М	i	87
		(2)			'	8	%	17	27			(₩ •	18	20	17	60
GPA and	Disposition by Scores	(T)	1.00-1.49	Below 10	Lo		12	9	7	1.50-1.99	Below 10	2			6	4

				:	• •	, ,		•	ř		,	. •	•	• •		•
	lt	(T2)	(4)	١	• (3	•	8	∾.	(1)		a k	•	-	4.2	•
964-65	2nd	Sell (14)	(4)	r-1		H			N	(6)	•	, a		c1	H	ယ
ET.		Sem (13)	(11)			ണ .	ജ	က	8	(6)	જા	H		H	-	4
	1	S [2]	(5)	 1	*	H	۲H	Н	· med	(3)		,	•	•		· rrfl
963-64	2nd		(13)		H	 N	.03	က	Ŋ	(3)	Н				·	ı
BI	15	Sem (10)	(2)	r-i	н	Н	¢	~	83	(5)			Н		83	GQ.
		S S (9)	(4)			–	r -1	-	H	(2)			· H	,	H	
962-63	To			H	83	гH	က	r-l	က	(2)	H		H		Н	es.
٠٢	4 L	Sen	(11)	~	က		8	m	ঝ	(5)	87				러 .	64 :
		S S (6)	(9)	Н	 4	က	~			(3)	mi				•	
961-62	2nd	Sem (5)	(15)	Н	83	Н	# 4	41	ဖ	(9)	႕	Ä	r-f		Н	63 .
	46	Sem (4)	(6)	н	m	Ḥ		က	က	(9)		•	6,	·	83	4
1961	1001	S S	(2)	H				, 		(1)	,		٠,	•		, a-l
15444	Enroll-	ment (2)		Ħ	\omega	18	20	17	67	1 () () () () () () () () () (. 00	8	20	6	5 0
	Office and	by Scores	2.00-2.49	Below 10	9		7	2	3 .	2.50-2.99	Below 10	10		75		

TABLE II-28 (continued)

		•	•		No.	• • •	•		<i>y</i> . ` .	
	S S (15)	(3)	••••••••••••••••••••••••••••••••••••••	più (。 ペ 私 イン	*	(1)		No.	H
964-55	2nd Sem (14)	(2)			 1	H	•			
F	Ist Sem (13)	(E)	•.	•		Ħ	4*			
	S S (12)	(1)	•	٠	~	•	,	•	,	· ·
963-64	TO EL	(3)	,		,	ಣ	••		•	
	Ist Sem (10)				•			•		
2	S S (9)	(1)				-	(1)	· · · · · · · · · · · · · · · · · · ·	•	-
962-63	Sem (8)	(2)	*		r-1,	Н			•	•
	Ist Sem	(2)		٣Î	н ,		.·	•		
,	S S (6)	;		• 2	:					•
9=195	Sem Sem (5)			• " •		•	(1)		~ -1	
	Sem (4)				~		(1)	;-1		
195	S S	3								
12.4.4	Enroll- ment		1 °	8	20	O		1 8	78	29
The second secon			Selow 10		~ ~	4	3.50-4.00	Below 10 10	73	? 네
ΔE	首首						67	8		

TABLE II-28 (continued)

*		•	Marie Carlos	
	ည က	12	£ 4 4	m
4-65	2nd Sem S	14)	E	4
.96T		3) (<u> </u>	
	Sem	무		,
-	83 83	(12)		
63-64	2nd. Sem			
196		(0)		٠.
	1st Sem			
	S	6		
962-63	2nd Sem	(8)		
6T	lst Sem			, K.,
2	လ လ	(9)		7 7 (4)
<u>961-62</u>	2nd Sem	(5)		
f	1st Sem	(4)		
<u>196</u>	\ \mathcal{O}	(3)		
ET I	ທ)		
1113	Enroll ment	(2)		29
E	Enro			
	ion			
and	Disposition Scores		Graduates Below 10 10 12	7
i G	Dist by S		677ac	
ૈસ છ	A Section of the second	P 5		3 G.

^{*}One student was given disciplinary dismissal.

Chapter III

ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18

The American College Test was administered to all students who entered The University of Mississippi during the summer and fall of 1961. Since there was no minimum score used for admission purposes, 103 students with scores of less than 15 were admitted. In Chapter II these students are analyzed in regard to schools, or college, in which they were enrolled and in regard to their grade point averages and disposition for a period of three years. This group made up 9.7 per cent of the total initial enrollment of 1062 students.

In this chapter, the students who entered the 1961-62 freshman class with composite standard scores of 15, 16, 17, 18 on the American College Test are analyzed in regard to the schools, or college, in which they were enrolled and in regard to their grade point averages and disposition for a period of four years and five summer sessions. This group had an initial enrollment of 250, or 23.5 per cent, of the total freshman class at the beginning of the first semester of 1961-62. This group was composed of approximately two and one half times as many students as the group which scored less than 15 on the American College Test.

Tables III-1 through III-8 present an analysis of the student enrollment on the basis of scores, by schools

for eight semesters. As was done in Chapter II, summer sessions are not included in enrollment data because of limited enrollments and limited changes which occur. Changes in schools which were made during the summer session may be noted in the semester following the summer session.

Tables III-10 through III-21 present the grade point averages and disposition of students through eight semesters and the summer sessions of 1961, 1962, 1963, 1964, and 1965.

Table III-22 presents a summary of the enrollments by schools through eight semesters.

Tables III-23 through III-24 present a summary of the grade point averages and disposition of the group under study.

TABLE III-1

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF FIRST SEMESTER

		College	or School	of Regist:	ration	
Standard Score	Liberal Arts	Bus. & Gov't.	Educa- tion	Engi- neering	Phar- macy	Total
(1)	(2)	(3)	(4)	(5)	(6)	(7)
15	18	. 16	13	1	· 3 · ·	51
16	21	24	12	2	4	63
17	38	15	9		1	63
18	37*	21*	12	1 ,	2	73*
Totals	114	76	46	4	10	250
Total Enrol ments in Schools	1- 5 74	229	126	103	30	1062
Per cent in each school making scor	· •					
of 15, 16, 17, 18	19.9	33.2	36.5	3.9	33.3	23.5

^{*}Two students with scores on the College Qualifying Test transferred to equivalent American College Test scores by use of percentiles

TABLE III-2

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF SECOND SEMESTER

Standard Score	Liberal Arts	College Bus. & Gov't.	Educa-	ol of Regist Ergi- neering	Phar- macy	Total
(1)	(2)	(3)	(4)	(5)	(6)	(7)
15	14	16	12	1	2	45
16	17	19	11		3	51.
17	32	17	7		1	57
18	28	21	10		2	61
Totals	91	73	40	 2	8	214

I. CLASSIFICATION BY SCHOOLS

At the beginning of the second year the enrollment for the initial group of 250 students who had made a composite standard score of 15, 16, 17, 18 on the American College Test in 1961 had dropped to 163, a total loss of 87 students, or 34.8 per cent, of the initial enrollment. Of the 114 students in the College of Liberal Arts at the beginning of the first semester of 1961-62, 73 remained at the opening of the third semester. Of these 73, eight had transferred to the School of Business and Government, 12 to the School of Education, and 53 remained in the College of Liberal Arts. Two students from the School of Education, and one student from the School of Education, and one student from the School of Engineering had transferred to the College of Liberal Arts. The College of Liberal Arts had lost 40, or 35.1 per cent, of its students from enrollment in the University after one year.

The School of Business and Government had an initial enrollment of 73. At the opening of the second year 53 remained in the same school, two transferred to the College of Liberal Arts, and two transferred to the School of Education. There were eight students from the College of Liberal Arts and one from the School of Engineering who had enrolled in the School of Business and Government. This school lost 17, or 23.3 per cent, of its initial enrollment from the University.

There were 46 students enrolled in the School of Education at the beginning of the first semester of 1961-62. At the opening of the third semester, four students from the

Arts and 21 remained in the School of Education. There were 12 students from the College of Liberal Arts and two from the School of Education at the School of Education at the beginning of the third semester. The loss from the University was 21, or 45.7 per cent, of the initial enrollment in the School of Education.

The School of Engineering retained only one of its initial enrollment and one transferred to the College of Liberal Arts. There was a loss of one, or 33 1/3 per cent, from the University of the beginning enrollment in the School of Engineering.

There were ten students in the group under study who initially enrolled in the School of Pharmacy. At the opening of the third semester, only five, or 50 per cent, of this group remained.

TABLE III-9

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF THIRD SEMESTER

Standard Score	Liberal Arts	College Bus & Gov't	or School Educa- tion	of Regist Engi- neering		Total
(U)	(2)	(3)	(4)	(5)	(6)	(7)
15	10	10	11	1	1	33
16	10	17	10		2	39
17	21	14	5		1	41
18	. 19	21	9		1	50
Totals	60	62	35	1	5	163

TABLE III-4

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF FOURTH SEMESTER

Standard Score	Liberal Arts	Bus. & Gov't.	or School Educa- tion	of Regist Engi- neering	ration Phar- macy	Total
(1)	(2)	(3)	(4)	(5)	(6)	(7)
15	8	8	10	1	1	28
16	8	14	10		2	34
17	. 12	12	7		1	32
18	18	19	9	1*	1	48
Totals	46	53	36	2*	5	142

^{*}One of the students previously enrolled in the School of Engineering returned to the University after an absence of two semesters

From the initial enrollment period at the beginning of the 1961-62 school year to the beginning of the third year of enrollment, the students who made composite standard scores of 15, 16, 17, 18 in 1961 dropped from 250 to 118, a loss of 132 students, or 52.8 per cent of the initial enrollment.

The drop from the beginning of the third senseter to the beginning of the fifth semester was from 163 to 118, a loss of 45 students, or 27.6 per cent, of the enrollment for the preceding year. Although some of the students in the initial enrollment had changed schools in the University, there were 30 of the initial 114 in the College of Liberal Arts, 36 of the original 76 in the School of Business and Government, 14 of the beginning 46 in the School of Education, two of the initial four in the School of Pharmacy remaining.

The transfers from one school to another were not as prevalent in this group as were the transfers among the students of the group which had scores of less than 15 on the American College Test. There were nine transfers to the School of Eduction from Business and Government and 13 to the School of Eduction from the initial enrollment in the College of Liberal Arts. The School of Business and Government had two students to transfer to the College of Liberal Arts and four to the School of Education after two years. There was one transfer to the College of Liberal Arts and one transfer to the School of Business and Government from the School of Education. One student who initially enrolled in the School of Engineering had previously



transferred to the College of Liberal Arts, but at the beginning of the fifth semester had enrolled in the School of Business and Government.

TABLE III-5

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF FIFTH SEMESTER

Standard Score	Liberal Arts	College Bus. & Gov't.	or School Educa- tion	of Regist:	Phar-	Total
(1)	(2)	(3)	(4)	neering (5)	macy (6)	(7)
15	5	7	9	1	1	23
16	. · · · 6	10	9		. 2	27
17		13	6		1	28
18	14	17	7	1	1	40
Totals	33	47	31	2.	5	118

TABLE III-6

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF SIXTH SEMESTER

Standard Score	Liberal Arts	College Bus. & Gov't.		of Regist: Engi- neering	ration Phar- macy	Total
(1)	(2)	(3)	(4)	(5)	(6)	(7)
15	5	5	9	1	1	21
16	6	12	9		2	29*
17	8	15	5		1	29*
18	15	17	8	1	1	42*
Totals	34*	49*	31*	2	5	121*

^{*}Eight students returned to The University of Mississippi after being out the previous semester, including two in the College of Liberal Arts, flour in the School of Business and Government, and two in the School of Education.

From an initial enrollment at the beginning of the 1961-62 school year to the beginning of the fourth year of enrollment in 1964-65, the students who made a composite standard score of 15, 16, 17, 18 in 1961 dropped from 250 to 89, a loss of 161 students, or a 64.4 per cent loss, from the initial enrollment. The loss between the beginning of the third year, or the fifth semester, and the beginning of the fourth year, or the seventh semester, was from 118 to 89, a loss of 29 students, or 24.6 per cent, of the enrollment for the previous year.

In the College of Liberal Arts there were 21 of the initial enrollment of 114. After the beginning of the fifth semester and at the beginning of the seventh semester one student transferred to the College of Liberal Arts from the School of Business and Government, the School of Education, and the School of Engineering, and two returned after an absence of a year.

After the beginning of the fifth semester and to the beginning of the seventh semester in the School of Business and Government, 31 of the initial enrollment of 76 remained, two transferred from the College of Liberal Arts, one transferred from the School of Education, and seven returned after an absence of one year. Two students transferred from the School of Business and Government to the College of Liberal Arts and one student transferred to the School of Education.

From the beginning of the fifth semester until the opening of the seventh semester, the School of Education retained 19 of its original enrollment of 46. One student transferred to the School of Education from the School of Business and Government and one returned after being out of school during the previous year. The School of Education lost one student to the College of Liberal Atta during the year.

The School of Engineering had an initial enrollment of four students which dropped to two students at the beginning of the fifth semester and at the beginning of the seventh semester there were no students with scores of 15, 16, 17, 18 left in the School of Engineering. One student transferred from the School of Engineering to the College of Liberal Arts.

In the School of Pharmacy there was an initial enrollment of ten students which dropped to five students at the beginning of the fifth semester and to three at the beginning of the seventh semester. Two students withdrew after the beginning of the fifth semester and prior to the opening of the seventh semester.

TABLE III-7

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI HEGINNING OF SEVENTH SEMESTER

Standard Score	Liberal Arts	Bus. & Gov't.	r School Educa- tion	of Registra Engi- neering	ation Phar- macy	Total
	(2)	(3)	(4)	(5)	(6)	(7)
15	2	6	5		1	14
16	5	9	5	•	. 1	23
17	6	13	8		1	24
18	13	11	4			26
Totals	26	39	21		3	89

TABLE III-8

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF EIGHTH SEMESTER

	7.7	College		of Regist:	ration
Standard Score	Liberal Arts	Bus. & Gov't.	Educa- tion	Engi- neering	Phar- Total macy
(L)	(2)	(3)	(4)	(5)	(6) (7)
15		4	3		
. 16	. 4	. 7	9,.		1 21
17	. 5	9	2		1 17
18	8	10			18
Totals	17	30	14		2 63

II. STUDENT ACHIEVEMENT

As indicated in Chapter II, the grade Point averages of students under study was based on a five point grading system when they were admitted to The University of Mississippi. At the beginning of the fifth semester of enrollment the grading system was changed to a four point grading system. All grades which are used in this study have been translated to the four point grading system by the formula shown on page 27.

Tables III-9 through III-12 show the annual cumulative grade point averages, the mean grade point average and the standard deviation for individual composite standard scores of 15, 16, 17, 18 on the American College Test, and the correlation coefficient between the cumulative grade point averages and the test scores.

Tables III-13 through III-25 indicate like data for each semester and summer session and the disposition of the students during each enrollment period.

TABLE III-9

CLANULATIVE GRADE POINT AVERAGES OF 1961-62 ENTER

NITH COMPOSITE STANDARD SCORES OF 15, 16, 17,

AMERICAN COLLEGE TEST, 1961-1962

THE UNIVERSITY OF MISSISSIPPI

	S. U.	CT.	67	2	.79		
	Wean CPA	(01)		69	1-62	92	
	3.50- 4.00	(6)					
		(8)		67	67	w	o
Average	2.50-	(2)	9	A	. •	o	28
Grade Point	2.00- 2.49	(9)	2	4	en H	12	51
-6r	1.50-	(2)	Ŋ	N.	9	4	0.9
	1.00- 1.49	(4)	9	, @	•	91	52
	Below 1.00	(3)			7		4 6
	2	(2)	9	G	29	2	24.3
	Standard Score			9		00	Totals

Summer sessions included.

TABLE ILI-10

CUMULATIVE GRADE POINT AVERAGES OF 1961-62 ENTER
WITH COMPOSITE STANDARD SCORES OF 15, 16, 17,
AMERICAN COLLEGE TEST, 1961-1963
THE UNIVERSITY OF WICET

				SI	ade Poini	. Average		The second secon		
Standard Score	Š	Below 1.00	1.00- 1.49	1.50-)- 2.00- 2.50-) 2.49 2.99	2.50 9.00	3.49-	3.50- 4.00	Mean	N G
	(2)	(3)	(4)	(5)	(9)	(7)	(8)	(6)	(10)	3
ja H	જ	07	Ŋ	2	o	41			1.95	54
9	့် ရာ		o	2	o	42			1.90	
	5	4	(0	16	10	4	63		о 0	ယ္
	9	•		` &	16		N		1.95	. 50
l'ota 1.s	159	ō	26	58	44	1.7	D			

TABLE III-11

CUMULATIVE GRADE POINT AVERAGES OF 1961-62 ENTERING F.
WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 19
THE UNIVERSET.

	Ä		20	27	က်	O	
	å	티	u)	ູ້	ທຸ	.50	
	an A	(10)	20	90	•06	9	
	Mean GPA		2.20	ญ่	Ň	2.06	8.
	-(el Marie Top Williams
	3.50. 4.00	6					
							La G Visi Visi
	.00- .49	(8)	r - 1	N	ଷ	က	ထ
	က က						
Average			5.F				
ver	2.50 2.99	E	ဖ	ဖ	വ	က	20
Point	2.00- 2.49	(9)	ာ	_	တ	œ	က
de .	w w					Ä	43
Grade	<u>.</u>						
	1.50 1.99	(5)	7	\omega	တ	က	3.7
	-00 -00 -00	4)	ଷ	မ	ell	œH	(0
	HH.						16
	3	11					
	Below 1.00	(3)			๗	H	സ
	o R	(2)	ည့	(O)	<u></u>	4. 23	127
							- v V
	건 			U.			'n
e ·	Standard Score		, D	9	B	0	Totals
	t g	#				- <u></u>	P

TABLE III-12

ING FRESHMEN 18 ON THE WITH COMPOSITE STANDARD SCORES OF 1961
WITH COMPOSITE STANDARD SCORES OF 15
AMERICAN CCLLEGE TEST, 1-9
THE UNIVERSITY OF MISSI

2.50-3.00-3.50-Hean 2.99 3.49 4.00 GPA (7) (8) (9) (10) 5 2.20 4 1 2.04 4 1 1.99 17 3					Gre	rade Point				,	• .
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Standard Score	No.	Below 1.00	1.00- 1.49	1.50-	2.49-		3,00- 3,49	3.50- 4.00	Hean GPA	S.D.
17 2 4 6 5 2.20 26 4 1 2.04 28 1 3 10 9 4 1 1.99 30 2 11 12 4 1 2.10 1s 101 1 1 3 2.10		(3)	(3)	(4)	(5)	(9)	(7)	(8)	(6)	(IIO)	(11)
26 4 1 2.04 28 1 3 10 9 4 1 1.99 30 2 11 12 4 1 2.10 1s 101 1 34 35 17 3	2	7		7	4	9	ស	•	B .	2,20	.52
28 1 3 10 9 4 1 1.99 2.10 3 2.10 3 15 17 3		88		(44)	တ	\omega	्च । ः	~	•	2.04	. 48
30 2 11 12 4 I 2-10. 15 101 1 11 34 35 17 3		28		က	10	. ග	4	ੰ ਜ	ď	1,99	.57
1s 101 1 11 34	œ	30		N	11	12	4	•⊢		2.10	46
	Totals	101	H		34	35	17	က	·	•	

TABLE III-13

FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST WHO WERE ENROLLED IN 1961 SUMMER SESSION

			#1	5	rade Poi	Grade Point Average	ige			
Standard	, S	Below 1.00	1.00- 1.49	1.50-	2.00- 2.49	2.50- 2.99	3.00- 3.49	3,50- 4,00	Kean	S.D.
(I)	(2)	(3)	(4)	(2)	(9)	(3)	(8)	(6)	(10)	
1 2	œ	 1	ന		က				1.56	•
91	10	H	H	ရာ	Ä	7	N		1.93	
4	လ	H		~	8		•	•	1.47	
8 .	&	•	H		က	83	m	r 	2.31	,
Totals	31	က	ໝ	S.	σ	5	က	Н		

Due to the small number of students enrolled during the summer of 1961, the standard deviations and the correlation have not been determined.

TABLE III-14

COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, THE UNIVERSITY OF MISSISSIPPI END OF FIRST SEMESTER

							Grade		Point Average	age			
dard	# .	With-	Dis-	Proba-	Below	1.00-	1.99-	$\mathbf{N} \mathbf{N}$	2.50- 2.99	00 49	3.50- 4.00	ㅁ	S.D.
acore Core	[2]	OLAWALS (3)	(4)	(5)	(9)	(2)		(6)	(10)		(12)	(13)	E C
15	co.	*-1	ည	Ĺ	ထ	ଷ	16	7	Ó			1.65	99*
9	6	•60	4	14	44	14	16	6	1	8		1.61	.72
9.54.5	60	H	.	10	13	15	. 12	15	쉭	red	8	1.64	.
; &	E	ઌ	, 10	19	14	10*	10	19	6	භ ₋	```.	1.64	.80
Totals	250	•	22	50	49	8 8	55	48	24	9	8		

TABLE III-15

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, THE UNIVERSITY OF MISSISSIPPI END OF SECOND SEMESTER

								これのファイン	4 + P - C				
dard		With- drawals	Dis- missals	Proba- tions	Below 1.00	ri ri	1.50- 1.99	2.00- 2.49	1-	3-3.00- 3-3.49	B 41	l CFU	S.D.
E	(2)	(3)	(季)	(2)	(9)	(2)	(8)	(6)	(10)	(77)	(12)	(123)	百
13	4 7	N	4	 1	က	တ	တ	4	4	—		1.88	.64
9	T T	~4	∞	4	9	12	10	10	œ	; H	,	1.74	.73
7	22		10	4	11	.	15	14	က	41	•	1:77	. 78
	6 1	H	7	ન	∞	п	ရ	. 15	7	ဖ		1,85	.77
Totals	4	4	29	13	28	43	47	ည	27	12	·	, <i>,</i>	**

TABLE III-16

POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, WHO ENROLLED IN THE 1962 SUMMER SESSION THE UNIVERSITY OF MISSISSIPPI

· · · · · · · · · · · · · · · · · · ·							AGE PO	Int Ave	rage			
Stan- dard		With-	Dist	Proba-	Below 1.00	1.50-	0- 2.00- 2.50- 3.00 9 2.49 2.99 3.49	2.50- 2.99	3.00- 3.49	3.50- 4.00	Mean	S.D.
SOOTE	Q S	CLAWALD	STOCKT		(4)	12	(6)		(11)	(77)	(13)	(#)
3	N.Z.	(5)	(#)		2		က	82		r - 4	2.18	88
က	o			•	က	ಞ	4	-	н	, ;	1.85	.56
6	64) F-1			(4)	_ත	, , ,	4		တ	ન `	2.01	T. 02
	9 G	+1	1 2	, ←	1 2	4	က	က	4	e-1 2	2.21	06
	n e	•	(က	*	4 8	ω	14	9	80	က		
STERRE	3									,		
	-											

TABLE III-17

POSITE STANDARD SCORES OF 15, 16, 17 COLLEGE TEST, THE UNIVERSITY OF M

				1 2			9	Grade Po	oint A	rerage	1		
dard fard		With-	Dis- missals	Proba-	Below 1.00	1.00-	1.50	2.00- 2.49	2.50- 2.99	0- 2.50- 3.00- 9 2.99 3.49	89 41	1991	S.D
	(2)	(3)	1	(5)	(9)	(<u>3</u>)	(8)	mi	(10)		(12)	(13)	(14)
(A)	က က	4	က	လ	ഹ	ဖ	Ŋ	10	r-I	63	•	1.67	-76
6	် ဗ	8	: 	4	ഗ	4	ET.	œ	4	~	• • •	1.85	.67
				, c 4	တ	ī.	 	10	4	1	H	1.63	.82
Q	9		73	S	न्त्र	⊢	14	14	_	7	Ÿ	1.93	09*
Wotals	163	o n `	14	16	20	26	43	42	16	ၒ	red	•	

TARIE TII-18

2 ENTERING FRESHMEN 18 ON THE AMERICAN SISSIPPI POINT AVERAGES AND DISPOSITION COMPOSITE STANDARD SCORES OF 15.

	S.D.	7	98*	5.33	.74	99.	
	Mean (13)	201	1.97	1.95	1,99	1-96	
	3.50-4.00	1447	* * * *	•	· ~	•	H
rerage	3 3.49	777	7	87	H	4	တ
Point A		757	7	က	ശ	ဖ	21
rade Po	2.00- 2.49	(2)	! ~	11	တ	11	38
[5]	1.50-2.00- 1.99 2.49	(8)	귁	12	ဆ	15	39
	00	(2)	87	ໜ	(ß	18
	Below 1.00	9	S	H	87	က	13
	Proba- tions	(2)	0	8	.64	က	o
	is- issals	(4)	വ		· ·	. N	O
	als m	· (S				~~	m
36.	With draw			4	ev	6 0	42
0.0	i. I. re No	(2					15 14
	dard Scor	B	12	E H	A		

TABLE III-19

AVERAGES AND DISPOSITION OF 1961
TE STANDARD SCORES OF 15, 16, 17
E TEST, WHO ENROLLED IN THE 1963
THE UNIVERSITY OF WISSIGN

•						,	
, ,	S.D.	THE STATE OF THE S	99.	.74	.70	88	ut Line in the second
	[444]	(13)	2.52	2.37	2.78	2, 28	
1	3.50- 4.00	(12)		H	N	H	ည
Average	3.00- 3.49	(TT)	· · ·	N	ക,	83	Φ
Point Av	2.50- 2.99	(10)	H	•	87	4	7.
Grade Po	2.00- 2.49	(6)	ଟ	m	8	H	
15	1.50-	(8)	ř - 1	ത	red	87	7
	1.00- 1.49	(2)	• •			H	H
	Below 7.00	(9)		•			-1
	Proba-	(5)		–		H	8
	Disa	(4)					
	With-	(3)	H				
		(2)	8		10	N	37
	dard		Ŋ	6			Tetals

83

TABLE III-20

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, THE UNIVERSITY OF MISSISSIPPI

1 2 2 2						Gr	Grade Po	Point A	Average	-		•
	With-	Dise	Proba-	Below 1.00	1.00-	1.50-	6 0	(O)	3.00- 3.49	3.50-	Mean	S.D.
		(4)	(5)	(9)	(7)		(6)	(10)		(12)		
		m	က		ເນ	ଷ	က	တ	± 4.	-	2.16	689
. >			8		M	ဖ	\omega	ွဲဖ	64	8	2.24	.71
or silver		64	က	—	, 네	41	œ	O	ဖ		2.17	69
		4	Ŋ	છ .	ဟ	4.; 9	12	<u></u>	ស	, , ,	2.13	.74
α		0	13	LO.	16	18	88	29	13	က		1

84

TABLE III-21

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, THE UNIVERSITY OF MISSISSIPPI

		S.D.	(14)	46	.76	8	.73	1
***	***	Maan	(13)		2.2%	2.24	2.34	
		3.50- 4.00	(12)		N	~	~	. 2 6
	ы	3.00 . 3.49	Œ	4	4	က	©	17
,	ı	50°			Ŋ	Ø	H	30
	ade Po	0- 2.00- 2.	(6)	ဖ		7	ଟ୍ୟ	31
	Gr	1.50- 1.99	(8)		~	့ ပ ွ	ഗ	19
		1.00- 1.49	(2)	rl	4	· বা	—	10
	4	Below	(9)		–		က	4
		Proba-	(5)		60	က	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	9
		Dis- missals	(4)		~	8	io,	1.0
The state of the s	The state of the s	With-	(3)	8				
का का का का			(8)	27	တ္	20	4	*
	0+52	dard	(T)	P	6		9	Wotals

Increase in enrollment for the sixth semester over the fifth semester caused by the return of seven students not enrolled during the fifth semester.

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN
WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18, ON THE AMERICAN
COLLEGE TEST, WHO ENROLLED IN THE 1964 SUMMER SESSION
THE UNIVERSITY OF MISSISSIPPI TABLE III-22

	S.D.	(14)	.73	86	98.	88	
	Mean	(13)	2.69	2.34	2.22	2.43	
S	3.50 4.00	(77)	8	8	N	H	7
Averages	3_00- 3_49		4	H	~	w	12
		(01)	r-1	က	'	၈	8
Grade Point	2.00- 2.49	(6)	∾	87	N	က	တ
63	1.50- 1.99	(8)	H	, ,	N		4
	1.00-	(3)		H	જા	 N	ည
	Below 1.00	(9)	H	~	H	က	7
	Proba-	(5)			.		5
	Dis-	(4)		8			
A TANK A	With-	(3)					67
		(2)	61) 		[ည်
	Stan-) • N		Totals

TABLE III-23

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, THE UNIVERSITY OF MISSISSIPPI END OF SEVENTH SEMESTER

- 1 mg - 1	, r			Tage .			
	S.D		w	. 82	7.	7.	
	-	(13)	2.34	2.13	2.16	2.32	
	3.5U-	(12)	, -	Ň	d	H	ည
erages	3.49 3.49		∾:	· — /	8	4	o
int Av	2.50- 2.99-		4.	4	ဖ	က	19
Grade Point Averages	2.00- 2.49	(6)	4	7	4	ဖ	24
Gr	1.99		r-d.	4	4	~	17
	N 1.00-	(2)		M	H	À	4
	Below 1.00	(9)	67	က	က	, , , , , , , , , , , , , , , , , , , 	ი
	Proba-	(5)		87	H		4
のなるのでは、大きのでは、大きのでは、大きのでは、大きのでは、大きのでは、大きのでは、大きのでは、大きのでは、大きのでは、大きのでは、大きのでは、大きのでは、大きのでは、大きのでは、大きのでは、大きのでは、	01.s-	(4)	67	က	ို စာ		o
の 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	With	Ordwalls (3)				0	7
		(S)	14	er C	, & 4	88	<u>ග</u>
A STATE		SCOTE				©	Totals

TABLE III-24

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, THE UNIVERSITY OF MISSISSIPPI END OF EIGHTH SEMESTER

O. F. C.							Ġ.	Grade Po	Point A	Averages			
dard	Z.	With-	Dist missels	Proba-	Below 1.00	1.00-	1.50- 1.99	2.00- 2.49	2.50- 2.99	3_00- 3_49	3.50- 4.00	Mean	S.D.
	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)	(13)	(14)
15	7			,		· .	r-l	87	8	H	H.	2.62	.75
9	21		8	H	~	' 27	Ħ	∞	က	9	•	2.36	.72
1	17	•	87	•		87	8	က	,	87	-	2.40	.72
18	18	e e e e e e e e e e e e e e e e e e e	, ન .	87		83	က	က	. 7	83	H	2.41	.62
Totals	63		ഹ	က	н	9	7	16	19	11	က		

TABLE III-25

1961-62 ENTERING FRESHMEN 17, 18 ON THE AMERICAN 1965 SUMMER SESSION GRADE POINT AVERAGES AND DISPOSITION OF WITH COMPOSITE STANDARD SCORES OF 15, 16, COLLEGE TEST, WHO ENROLLED IN THE THE UNIVERSITY OF MISSI

	S.D.		[#·	.47	.71	.84	
	Mean 3		2.33	2.40	2.16	2.30	
	3.50-4.00			· / -		.	m
DVATARAS	3.00- 3.49	1777	Н	Н	વ્યા	•	9
Day ht Au	'lo or	777		. 63	.	1	 ♣1
1	2.49 2.49	(2)	7	83	4,	4	12
7	5 _	(0)		H	က		4
	1.49	3			~	~	2
	Below 1.00	(0)			H		H
	Proba- tions	(2)		Н			1
	Dis- missals	(4)			Н		1 4
	With- drawals	(3)					
	No.	(2)	က	9	15	ဖ	30
	Stan- dard Score	(E)	15	16	17	18	Totals

III. SUMMARY OF REPORTS

In the preceding tables of this chapter the data indicate enrollments in schools by semester and by scores on the American College Test and the grade point averages and disposition of the entering freshmen of 1961-62 with scores on the American College Test of 15, 16, 17, 18.

The enrollments indicate a drop of students with scores on the test of 15, 16, 17, 18 from an initial enrollment of 250 during the first semester of 1961-62 to 63 at the end of the second semester of 1964-65, or the end of eight semesters. The correlation coefficients of the grade point averages varied from .096 for 154 students earning grades during the third semester to -.070 for 36 students earning grades during the 1963 summer session indicating practically no correlations between grade point averages and ACT scores in the range of 15, 16, 17, 18.

In Table II-26 the enrollments by schools have been summarized by schools and by semesters.

Table III-27 shows a summary of the grade point averages and disposition, by semesters and summer sessions, of the 1961-62 entering freshmen with composite standard scores of 15, 16, 17, 18 on the American College Test.

Table III-28 shows a complete summary of the grade point averages and disposition, by semesters and summer sessions, for individual scores of the 1961-62 entering freshmen with composite standard scores of 15, 16, 17, 18 on the American College Test.

TABLE III-26

ENROLLMENT OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, THE UNIVERSITY OF MISSISSIPPI 1961-62----1964-65

·	<u> </u>		Ye	ar of	Enroll	ment		
	196	1-62		1962-63		1963-64		4-65
School or College	lst Sem.	2nd Sem.	lst Sem.	2nd Sem.	lst Sem.		lst Sem.	2nd Sem.
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Liberal Arts	114	91,	60	4 6	33	34	26	17
Business and Government	76	73	62	53	47	49	39	30
Education	46	40	35	36	31	31	21	14
Engineering	4	2	1	2	2	2		•
Pharmacy	10	8	5	5	5	5	3	2
Totals	25 0	214	163	142	118	121*	89	63

^{*}The increase in enrollment for the second semester of 1963-64 over the first semester of the same year caused by return of seven students not enrolled during the first semester

TABLE III-27

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, 18 ON THE AMERICAN COLLEGE TEST, THE UNIVERSITY OF MISSISSIPPI 1961-62---1964-65

(12) (13) (14)

163 142
214 53
(D)

*Additional students enrolled who had not been during preceding semester.

91

TABLE III-28

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF 15, 16, 17, THE AMERICAN COLLEGE TEST, BY TEST SCORES, P. BY SEMESTERS, THE UNIVERSITY OF MISSISSIPPI 1961-62---1964-65

egh and	Initial			1961-62			1962-63	8	T	963-64			964-65	2
Disposition by Scores	Enroll-ment	M	lst Sem	2nd Sem	S S	lst Sem	2nd Sem	1 00	1st Sem	2nd Sem		1st Sem	2nd Sem	လ လ
(1)	(2)	വ	(4)	(5)	ICC 33	(2)	(8)	ЮП	(10)	(E)	Married II	(13)	(14)	
Enrollments:		(31)	(250) (214)	(214)	(53)	(163)	(163)(142)	(32)	(118)	(121*	(121*)(53)	(83)	(63)	(63) (89)
15	E,	&	51	45	6	ဗ	28	Φ	23	21	12	, जी । ान	4	က
9[63	10	ဗ	21	12	68	34	7	27	29	13	23	21	w
17.	63	လ	63	22	13	41	32	10	28	_\ 8	12	24	17	75
* ***	73		73	61	19	20	4 4	12	40	42	16	83 	18	9
Withdrawals:		· · · · · · · · · · · · · · · · · · ·	8	(4)	(2)	(6)	(3)	(1)	(1)	(4)	(2)	(2)		0
9	51		~	2	. ન ે	4	r-1 :	;;	, ci ,	Ø	,			The state of the s
9 T	63		က	ائم		භ				,	#	•	«.	
17	89		H			Ħ,	. •	:	*	-	•	5. *		1. The state of th
87	73		က	-	 1	–	8			H		N		

ERIC

TABLE III-28 (continued)

	83		(1)			ન		3		H	3 47	: .:
199	2nd Sem	(T.F.)	(5)		N	6	H	3	· · · · · · · · · · · · · · · · · · ·	—	/ 	~
19	1st Sem	£101	6	M **.	် က် န	•		(3)		Ö	~ (i)	· H (
	l rat		(2)	H	• (4)	H	6 0	(2)		e alla 1	~	
963-64	Sem	(77)	(10)	ent :	83	63	က်	9	•	က်	က	
6	71	(32)	(10)	က	,	%	ক	(13)	က	8	က	က
	က လ	(6)	. *	*		. <u>.</u>	•	(2)		-		H
069_63	2nd Sem	(8)	(6)	ũ	~	H.	8	(6)	N	63	83	က
	1st Sem	(2)	(14)	က	-1	&	83	(16)	က	41	8	ស
	ω ω	(9)	(3)		\	H	87	(4)			က	H
2 6 2 4	Znd Znd Sen	II	(53)	41	ထ .	10	4	(13)	—	4	4	4
,	1st Sem	(4)	(22)	.	4	œ	ູນ	(20)	7	14	10	19
	1361 S S	16.4 18										
	Initial Enroll-	(2)	Andrew State of the Control of the C	51	63	. 63	73	•	27	89	မ	73
	GPA and Disposition		Dismissals:	15	9	Å	8	Probations:	12	9	17	8

TABLE III-28 (continued)

	• •	- •				K			1	105-6-201		D.	1964-65		5
SPA and Disposition by Scores	Initial Enroll- ment	1961 S S		1961-62 2nd n Sem	SS S	19 Ist Sem	2nd Sem S	88	Sei Sei	Sen Sen	S S [2]	18t	∆ € 4	S S (15)	
3	(2)	(3)	. (4)	(6)	9	3		5			\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	,		ý ,	
Grade Point								· ·	ı	Section 1		•			, .
Below 1.00		(3)	(49)	(28)	(4)	(20)	(13)	(1)	(2)	(4)	(2)	(6)	(1)	(1)	7
10.	21	H	œ	က		ß	വ				~	N			4.
16	89	1 — 1 1 — — 1 12 — 22,	14	ဖ		۰ ب	Н		r-1	H	~	က် က ်	ਜ਼ੇ	m a Car	port 🐣
6	89	Н	13	11	က	တ	8	,	Н		# ~	****	•	ed .	£
881	73		14	∞	H	—	ស	pad .	က	က	ආ .	-			, ,
.00-1-49		(5)	(58)	(43)	(8)	(36)	(18)	(1)	(16)	(10)	(5)	3	9		
, S	21	ဓာ	12	ග	67	ပ	87		ريا د	rH		3	* 7 J		b.
9	6 3	H	14	15	က	4	Ŋ		8	4	m	A	N	2 A	
17	63		12	∞	H	ທ	9		4	4	77	H	7	, ,,,	
18	.73	÷	17	11	87	H	വ	Н	ູນ	H	67	e	A		ty.
-		•												•	

ERIC

TABLE III-28 (continued)

		. 业量 -			*			•			.50
S	(15)	7		H	.	j.	(12)	64	8	4	₹
2nd Sem	F E	(2)	H		N	ഩ	(31)	~	©	ෆ	ಣ
1st Sem	(13)	(11)	H	4	4	.	(24)	4	7	À	Ø
1		(4)	4	m	67		(6)	7	?? 7 3	83	က
		(13)		.	ဖ	က	(31)	ထ	ເນ		L
i I	(10)	(18)	7	ဖ	₹.	ဖ	(33)	လ	, დ ,	∞ ∺	12
Ŋ	io i	(7)	H	က	. 	82	2	رر. ر من	· —	02	-
2nd Sem	(8)	(38)	4	12	, დ	12	(38)	~	Ħ	တ	
	(2)	(43)	ß	13	11	4.1	(42)	10	&	10	4
l v	KOL	8		න \	H	4	(14)	မ်ာ	्य ्या ः	4	က
2nd Sem	(2)	(47)	တ	10	12	13	(53)	14	10	14	15
Ist	(4)	(55)	16	16	13	10	(48)	7	7	15	19
S. S.	(3)	(5)		က	2		(6)	က	r-1	, N	ന ്
Enroll-	(2)		21	မှ	63	2	A)	2	e e e e e e e e e e e e e e e e e e e	ဗ္ဗ	2
r in	1 1	1.50-1.99	. 15	9	47	•	2.00-2.49	1	9.		
	Solution Enroll Ist 2nd 1st 2n	XOSITION Enroll-Enroll-Enroll-Strong Sem	Scores Inst 2nd 1st 2nd 1st 2nd 1st 2nd 1st 2nd Scores Ment S S Sem Sem	Sosition Enroll Enroll Enroll S S Sem Sem Sem Sem Sem Sem Sem Sem Sem	Solution Enroll Ist 2nd 1st 2nd 1st 2nd 1st 2nd Sem S Sem	Osition Enroll Ist 2nd	Solution Enroll-Siger Ist 2nd Sem	Serition Enroll-Enroll-Enroll-Enroll-SS Sem Sem Sem Sem Sem Sem Sem Sem Sem S	Second Sear Sear Sear Sear Sear Sear Sear Sear	Self-tion Entrollaries Ist Snew Sem	Osition Enroll Enroll Ist 2nd Sem

TABLE III-28 (continued)

E .	· · · · · · · · ·	₹ a	æ	5.5		- 6 ₁						· .		
1	=	S	(27)	(4)	1	M	N		(9)	H	Ħ	41	N	
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	964-65 2nd	1		(19)	~	m	-	Tight .	3	H	.	W	N	
			(113)	(13)	4		0	W	(6)	€¶	Ä.	ุณ	4	
		SS	[77]	(8)		က	-	က	(12)	ঝ ,	~	M	ഹ	
	963-64 2nd	He He		(30)	&	, w	~.0	Ħ	(17)	ঝ	ঝ	က	ပ	1
	13		(07)	(29)	ာ	ဟ ့	'n	O .	(113)	•	6 7	ဖ	ທຸ	
	,-	S	(A)	(1)	H		61	4	(8)	—	~	က	~	
	962-63 2nd	Sem	(<u>x</u>)	(21)	2	်က	'n	φ	(6)	.2	7	H	4	
	1st	Sem		(16)	H.	ď	ঝ	7	(9)	6.4	C)		N	
			(9)	(9)	8	-	**************************************	က	(8)		H	က	4	
3	961-62 2nd	Sem	(5)	(27)	_	œ	`	F .6	(12)	·	H	4	ဖ	
		Sem	(4)	(24)	ဖ		4	7	(9)		7 3		က	
	1961	SS	(8)	(2)		୍ଷ	o a a a a a a a a a a a a a a a a a a a	~	(3)		87		\	n de
M-7	Lial 511+		7									× ×	د بردید، د بردید، د بید	
	Initia Enroll	ment		16.00	5	63	ဗွဲ့	73		21	93	63	73	
一十二十二十二十二十二十二十二十二十二十二十二十二十二十二十二十二十二十二十二	1 Lion	res		2,99		X	A MARINE	ton 1 Co	0					
	GPA and Disposi	by Scores		50	L	9	A	8	3.00	H 2	.	5	, <u>æ</u>	
≟n'	io U	ᆈ.	F	No.	\$ 100 h		a lunk a		က	₩	,		0	1))))

TABLE III-28 (continued)

gen a di				. , , , , , ,	۵	# .	r e fæje. Gr		±.	ي د يو د يو دو. دي د يو د يو دو		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
. S	[15]	3					(8)		က	40	H	
964–65 2nd Sem	(14)	(3)	m			–	(34)	S	æ		Ä	, , , , , , , , , , , , , , , , , , ,
1-4	(13)	(2)	(8			(10)	7	N		b	
מז ו	(12)	(2)	, cd	e 3	જ	H	3				N	
963-64 2nd Sem 3	(11)	(9)		. ~	. 😝	•	(2)	H				
	(10)	(3)	1	0								
ω Ω	6)	3	,		∾.	-1						
962-63 2nd Sem	(8)	(1)										
	(2)	(1)	1	*	1							
1 1 021	(9)	(8)	1		H		, 1					
. 1961 - 62 E 2nd n Sem	(5)											
1st Sem	(4)	(2)			~							
1 <u>1961</u> S S	(8)	Ĵ										
Initial Enroll- ment	2	No.	-\- -\-	္မွ	္မွာ	73			က	m	~	
			S	9	©			S	63	6	2	
d ition res		0					. <u>1</u>		V.			
GPA and Disposition by Scores	3	3.50-4.00	19	"9T"	17	.œ' ₂r⊒	Graduates	, L	9 T	2	. T3	100

*The increase in enrollment for the second semester of 1963-64 over the first semester of the same year came about because of the return of seven students who were not enrolled during the first semester.

Chapter IV

ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18

The third group of students under study includes all students who enrolled in the 1961-62 entering freshman class with composite standard scores of greater than 18 on the American College Test. This group was composed of 709 students which comprised 66.8 per cent of the total enrollment. The highest score attained by a student in this particular class at The University of Mississippi was 32 out of a possible score of 36.

Tables IV-1 through IV-8 indicate the enrollments by schools for freshmen who were included in this group for each semester. As in Chapter II and Chapter III, the summer sessions have not been included because of the limited number of students enrolled. Changes in student classifications which were made during summer school are noted in the data presented for the semester following the summer session.

Tables IV-9 through IV-12 show the cumulative grade point averages prepared for annual periods for the preceding time of enrollment.

Tables IV-13 through IV-25 indicate the performance and disposition, by semester and summer sessions, and by ACT scores, of the 709 students who entered the 1961-62 freshman class with composite standard scores of greater than 18.

Table IV-28 summarizes the envollments in each school, or college, based on semesters through the entire period of envollment at The University of Mississippi.

TABLE IV-1

INITIAL ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF FIRST SEMESTER

		College				
Standard Secre	Liberal	Bus. &	Educa-	Engi-	Phar-	m_4_7
Score (1)	Arts (2)	Gov't. (3)	tion (4)	neering (5)	macy (6)	Total
19	5.7	29	18	2	2	108
20	44	23	9	6	2	84
21	52	18	.7	4	1	82
22	5 7	19	3	11	9	99
23	55	Õ	2	9		75
24	47	6	1	10		64
25	27	11	4	13	1	56
26	27	3	2	12	1	45
27	21	· 2	3	10	· 1	37
28	16	1	•	5		22
29	9			4		13
3 0	9			3		12
31	. 5	2		1.		8
32	. . 2			2		, 4
otals	428	123	49	92	17	709
otal enro	11-			• • • • • • • • • • • • • • • • • • • •		
ents in chools	574	229	126	103	30	1062
Per cent is chool make cores of	cing .			erit Karangan dan Karangan		•
han 18	74.6	53.7	38.9	89.3	56.7	66.

TABLE IV-2

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF SECOND SEMESTER

		College	or Schoo	l of Regis	tration	1
Standard Score	Liberal Arts	Bus. & Gov't.	Educa- tion	Engi	Phar-	
(1)	(2)	(3)	(4)	neering	macy (6)	Total
19	45	29	19	1	2	96
2 0	38	23	12	4	1	7 8
21	47	20	6	4	1	78
22	51	23	5	6	8	93
23	52	12	3	4		71
24	41	11	2	7	•	61
25	25	11	4	9	2	51
26	25	3	1	11	1	41
27	20	2 ,	3	8	· ,1	34
28	17	. 18 ·		4	,	22
29	9			4		13
30	8	•		3	1	12
31	6	1	•	1	•	8
32	2			2	1	4
Totals	386	136	55	68	17	662

I. CLASSIFICATION E" SCHOOLS

From Table IV-3, it can be seen that the enrollment of the group under study had dropped from 709 as the initial enrollment to 525 at the beginning of the second year. represents a loss of 30.0 per cent of the initial group. There were 428 students enrolled in the College of Liberal Arts at the beginning of the first semester of 1961-62 and 286 remained at the beginning of the third semester. 428 students enrolled in the College of Liberal Arts at the beginning of the first semester of 1961-62, 310 remained in the University at the beginning of the second year, or the third semester. Of the 310 students remaining, 265 continued in the College of Liberal Arts, 22 transferred to the School of Business and Government, 19 to the School of Education, one to the School of Engineering, and three to the School of Pharmacy. One hundred eighteen students, or 27.6 per cent, who were initially enrolled in the College of Liberal Arts had dropped out completely.

In the School of Business and Government there was an initial enrollment of 123. At the beginning of the third semester, 89 continued in the University with 83 continuing in the School of Business and Government, four transferring to the College of Liberal Arts, and two to the School of Education. Thirty-four, or 27.6 per cent, had withdrawn from the University.

There were 49 students in the School of Education at the beginning of the session 1961-62. Of these, 31 continued in the School of Education, two changed to the College of Liberal Arts, and one to the School of Pharmacy. Fifteen, or 30.6 per cent, were not enrolled at the beginning of the second year.

The School of Engineering had an initial enrollment of 92 students from the group with scores of greater than 18 on the American College Test. At the beginning of the second year 47 remained in the School of Engineering, 15 transferred to the College of Liberal Arts, 12 to the School of Business and Government, and two to the School of Education. Fifteen, or 16.3 per cent, of the initial enrollment had dropped out at the beginning of the second year.

The initial enrollment in the School of Pharmacy was 17.

The second year began with 15 continuing in the School of

Pharmacy and one transferring to the School of Business and

Government. One, or 5.9 per cent, of the original enroll
ment had withdrawn from the University.

TABLE IV-3

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF THIRD SEMESTER

			or School	of Regis	tration	
Standard	Liberal	Bus. &	Educa-	Engi-	Phar-	
Score .	Arts	Gov't.	tion	neering	macy	Total
(1)	(2)	(3)	(4)	(5)	(6)	(7)
19	28	22	17	1	2 .	7 0
20	21	20	12	1	1	55
21	32	1 20	8	3	2	65
22	36	∖20	6	5	7	74
95	36	10	4	3		53
24	33	9	1 2 2	5	1	49
25	22	9	3	6	2	42
26	23	3	1	7	1	35
27	19	4	2	.5	. 1	31
28	14	2		4	•	20
29	7	•		4		11
30	8	•		2	1	11
31	4			1		5
32	3		. · · · · · · · · · · · · · · · · · · ·	1	N.	4
Totals	286	119	55	48	17	525

TABLE IV-4

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF FOURTH SEMESTER

		College			ration	
Standard Score	Liberal Arts	Bus. & Gov't.	Educa- tion	Engi-	Phar-	
(1)	(2)	(3)	(4)	neering (5)	macy (6)	Total
19	22	22	18	1	2	65
20	17	18	11	2	1	49
21	27	20	8	3	4 2	60
22	31	21	7	4	6	69
23	31	11	4	2	· .	48
24	28	9	3	3		43
25	19	8	2	5	3	37
26	20	5	1	3		29
27	19	3	4	3	1	3 0
28	13	2		4		19
29	8			3	•	11
30	6	1	A Company of the Comp	2	1	10
31	4	ii. Le iii.		1		5
32	2	•		1	•	3
Totals	247	120	58	37	16	478

As indicated in Table IV-5 the enrollment of the group of students under study in this chapter dropped from an initial enrollment of 709 to 421 at the beginning of the third year, a loss of 288 students, or 40.6 per cent.

Enrollment in the College of Liberal Arts in the beginn ming of the second year was 286, of which 220 remained at the University at the beginning of the third year. This represented a loss of 23.1 percent of the previous year's enrollment. At the beginning of the third year the enrollment of the College of Liberal Arts was 203 which included 192 students who continued in Liberal Arts, five who transferred from the School of Business and Government, and six who transferred from the School of Engineering. There were 28 students who were enrolled in the College of Liberal Arts during the second year who transferred to other schools as follows: School of Business and Government, 15; School of Education, 11; and School of Pharmacy, two.

Of the 119 students who were enrolled in the School of Business and Government at the beginning of the second year, 105, or 18.2 per cent, remained in the University at the beginning of the third year. There were 119 students who enrolled in the School of Business and Government at the beginning of the third year. This enrollment consisted of 100 students who were enrolled in the school af the beginning of the previous year, 15 who transferred from the College of Liberal Arts, three who transferred from the School of Engineering, and one who transferred from the School of Pharmacy. Of the previous year's enrollment in Business and Government

there were five students who transferred to the College of Liberal Arts.

In the School of Education the second year enrollment was 55. Forty-three, or 78.2 per cent, of these were enrolled in the University at the beginning of the third year. All of these students continued in the School of Education. There were 11 students who transferred to the School of Education from the College of Liberal Arts after the opening of the previous year.

Of the 48 students who were enrolled in the School of Engineering at the beginning of the second year, 36, or 75.0 per cent, remained in the University at the beginning of the third year. At the beginning of the third year of enrollment the School of Engineering had 27 students enrolled who had been enrolled in the school during the previous year. Of the other students who had been enrolled in Engineering during the previous year and were still enrolled in the University, six transferred to the College of Liberal Arts and three transferred to the School of Business and Government.

The entire group of 17 students were enrolled in the School of Pharmacy at the beginning of the third year. All except one student continued in the School of Pharmacy. This one student transferred to the School of Business and Government. There were two students who transferred to the School of Pharmacy from the College of Liberal Arts after the opening of the second year making a total of 18.

TABLE IV-5

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF FIFTH SEMESTER

		College	or School	of Regist	tration	
Standard Score	Liberal Arts	Bus. & Gov't.	Educa- tion	Engi- neering	Phar-	
(1)	(2)	(3)	(4)	(5)	macy (6)	<u> </u>
19	14	22	18	1	2	57
20	15	15	11		1	42
21	21	17	8	1	2	49
22	27	21	5	3	8	64
23	24	12	3			39
24	24	11	3	2		40
25	17	9	3	5	3	37
26	15	5	10 10 100	4		24
27	16	. 4	3	2	1	26
28	13	2		3		18
29	7			2		9
30	4	1		2	1	8
31	4			1		5
32	2			1		3
Totals	203	119	54	27	18	421

TABLE IV-6

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF SIXTH SEMESTER

		College	or Schoo	l of Regis	tration	
Standard Score	Liberal	Bus. &	Educa-	Engi-	Phar-	., ,
(1)	Arts (2)	Gov't. (3)	tion (4)	neering (5)	macy (6)	Total
The state of the s			1 2		(0)	(7)
19	14	20	17	1	2	54
20	13	14	10		1	38
21	50	16	ò			
	20	10	9		· \ 2	48
22	24	22	5	3	9	63
23	24	12	3		The second secon	39
24		10				
23	22	TO TO	4	2		38
25	17	9	4	5	3.	38
26	15	5		$oldsymbol{4}$	-	24
27		4				
4/	15	4	3	2		25
28	11 .	2	The second second	3		16
29	7	1 HAT THE		2		9
				The state of the s		
30	4			2	1	7
31	4					5
32	2					3
					10	3
l otals	192	114	55	27	19	407
					TJ	#U/

As Table IV-7 indicates, the enrollment of students who entered the 1961-62 freshman class with composite stnadard scores of greater than 18 had dropped from an initial count of 709 to 325 at the beginning of the fourth year of enrollment. This represents a loss of 347 students, or 48.9 per cent, of the initial enrollment when the 37 who graduated previously are not included in the losses.

The College of Liberal Arts, which had an enrollment of 203 students at the beginning of the third year of enrollment, began the fourth year with 138 students, or 62.7 per cent, of the enrollment for the previous year. Of the 138 students still enrolled in the University from the group which was enrolled in the College of Liberal Arts at the beginning of the third year of enrollment, 127 remained in the College of Liberal Arts at the beginning of the fourth year, six transferred to the School of Business and Government, three to the School of Education, and one to the School of Pharmacy. There were two students who returned to the University in the College of Liberal Arts after an absence of one year and four who returned to the same College after an absence of two years. One student previously enrolled in the School of Engineering returned after two years absence and enrolled in the College of Liberal Arts. transfers to the College of Liberal Arts of two students from the School of Business and Government, one student from the School of Education, and one student from the School of Pharmacy.

There were 99 students, or 83.2 per cent, of the previous year's enrollment of 119 students who were in the
School of Business and Government, who remained in the University at the beginning of the fourth year. Mety-five of
these students who were enrolled in the School of Business
and Government at the beginning of the previous year remained
in the same school at the beginning of the fourth year. Of
the students who were enrolled in the School of Business and
Government at the beginning of the third year, two transferred to the College of Liberal Arts and two transferred to
the School of Education. One student, enrolled in the School
of Business and Government previously, returned after an
absence of two years. Six students transferred from the College of Liberal Arts to the School of Business and Government.

In the School of Education, there were 37, or 68.5 per cent, of the previous year's enrollment of 54 remaining in the University. Of the 37 students remaining who were enrolled in the School of Education at the beginning of the third year, 36 remained in the School of Education at the beginning of the fourth year and one transferred to the College of Liberal Arts. There was one student, who had been enrolled in the School of Business and Government previously, who returned to the University in the School of Education after an absence of two years and one student, who had been enrolled in the College of Liberal Arts previously who returned to the University in the School of Education after an absence of one year.

Three students from the College of Liberal Arts and two students from the School of Business and Government trans-

ferred to the School of Education making a total of 48.

The School of Engineering enrolled 27 students at the beginning of the third year of which 24, or 88.9 per cent. remained in the University at the opening of the third year. These 24 students were still enrolled in the School of Engineering at the beginning of the fourth year.

Of the 18 students who were enrolled at the beginning of the third year in the School of Pharmacy, 17, or 94.4 per cent, were enrolled in the University at the beginning of the fourth year. One student had transferred to the College of Liberal Arts since the beginning of the third year, while a single student transferred from the College of Liberal Arts to the School of Pharmacy. One student, who had been enrolled in the College of Liberal Arts previously, returned to the University after an absence of one year and enrolled in the School of Pharmacy at the beginning of the fourth year.

TABLE IV-7

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF SEVENTH SEMESTER

		College	or School	of Regis	ration	
Standard	Liberal	Bus. &	Educa-	Engi-	Phar-	
Score	Arts	Gov't.	fion-	neering	. macy	Total
(1)	(2)	(3)	(4)	(5)	(6)	(7)
19	11	18	10	1 -	3	43
20	10	12	8		• 1	31
21	16	15	5		1	37
22	15	18	6	3	7 -	49
23	18	11	3		1	.33
24	16	g.	5	2	•	32
25	13	7	2	4	. 3	29
26	9	5	1	3		18
27	. 10	5	.3	2	1	21
28	9	2		3	•	14
. 29 .	. 3	4	· di	. 2		, 5
30.	4			2	1	7
31	3			1		4
32	1			1	••	2
Totals	138	102	43	24	18	325

TABLE IV-8

ENROLLMENT OF THE 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST, BY SCHOOLS, THE UNIVERSITY OF MISSISSIPPI BEGINNING OF EIGHTH SEMESTER

		College	or School	ol of Regis	ration	
Standard	Liberal	Bus. &	Educa-	Engi-	Phar-	
Score	Arts	Gov't.	tion	neering	macy	Total
(1)	(2)	.(3) .	-(4)	(5)	(6)	(7)
19	8	17	8	. 1	. 2	36、
20	9	9	5		. 1	24
21	12	10	i .		i	24
22	10	17	4	1	5	37
-23	. 10	11	1		1	23
24	15	8	2	1		26
.25	. 12.	6	1	2	: 3	24
26	6	5	V-	4		15
27	. 10	3	. .	1	1	18
28	6	1		2		9
. 29	1			2		3
30	4	of spring is the spring to the spring is		. 1	1.	6
31	3			1		4
32	1			1	.•	2
Totals	107	87	25	17	15	251

II. Student Achievement

As in Chapter II and Chapter III the grades of the entering freshmen of 1961-62 have been translated to the four point grading system by use of the formula which is given on page 27.

grade point averages, the mean grade point average and the standard deviation for individual composite standard scores of greater than 18 on the American College Test, and the correlation coefficient between the cumulative grade point averages and the ACT scores. Tables IV-13 through IV-25 indicate like data for each semester and summer session and the disposition of the students during each enrollment period.

TABLE IV-9

CUMULATIVE GRADE POINT AVERAGES OF 1961-62 ENTERI

	n S	国	1 000000000000000000000000000000000000	
	Wean GPA	010	4 - 4 4 4 4 4 4 4 4 6 4 6 6 6 6 6 6 6 6	
	3.50- 4.00	(8)	. မာ <i>က</i> မာ	5.5
	3.00- 3.49	(8)	4@QUCP@@D@@WAH	8
•	Average 2.50- 2.99	(7)	2112111 11702718818141	150
e e	de Point 2.00- 2.49	(9)	ФФФСФФПН МППМППН	176
	Grade 1.50-2 1.99 2	(5)	MHHHH NYOOHOOMOH H	120
	1.49	(季)	エューム・アのの41	Ż.
	Below 1.00	(8)	MO 4.F.4.WHH	4 2
	No.	(2)	ロ ロ の の の の の の の の の の の の の	707
	Standard Score		122222222222222 201224707220122	Totals

Note: Summer sessions included.

CUMULATIVE GRADE POINT AVERAGES OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE THE UNIVERSITY OF WITH 1963

		Ö,		らなるのでではならばままだ。	
• 4		Mean		2 4 4 4 4 4 4 4 6 6 6 6 6 6 6 6 6 6 6 6	
		3.50 4.06	(6)	—— °ою 4044040000	07
		3.49	(8)	ಬಬಹ ಇರಿ ದಿರ4ಜಲ4ಭರಗ	74
	-		(2)	00000000000000000000000000000000000000	117
	de Point	2.00- 2.49	(9)	2 2 2 2 1 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2	132
	Grade	1.50 1.99	(5)	4444 808460768 4	108
÷.		_00- 48	(4)		4.
		Ветом.	(3)	MMVH MHH	14
		0	(2)	で写像でいきまままなまれる。	52 30
		Standard Score	3	14444444444444444444444444444444444444	Totals

CUMULATIVE GRADE POINT AVERAGES OF 1961-62 ENTERING FRESHMEN
WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE
THE UNIVERSITY OF 1961-1964

	d g		.51	ပ် က လ တ ထ	(a) (b) (c)	က တက် လူက လူ	N (0, N)		
	Mean GPA	(01)			医囊部侧侧侧	2 2 8 2 8 3 2 3 2 3 3 3 3 3 3 3 3 3 3 3 3 3 3			
	3.50 4.00	(6)	HH	. m m	400	က ဟ တ	нαн		원
C	3 00 3 49	(8)	ri o o	၀ဌ က	တ ဥ	(O 4) (O)	м м ч		7 6
t Average	2.50-	(7)	∞ ⊢7		ru⊗ 4	ក ក	60		118
Je	2.00 2.49	(9)	17	2 4 4 10 4 4	დ 4 დ	44			118
Gra	_ G G G	(.5)		၁ တ (၂၁) 1 ၂	ω 4 Φ	NWH			79
	- 0 7 0 4 - 1	(4)	∞	H eo es	n -ii				21
	Below 1.00	(2)	N						7
	څ	(2)	145 135	8.4 8.1	၁ တ မာ	N ⊟ N ⊗ の (ထိ ဟု က		439
	Standard Score	(3)	200	1000 1000	4 2 2 4 3 0	C & O	D -1 & €) :	Totals

COMULATIVE CRADE POINT AVERAGES OF 1961-62 ENTERING FRES
WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON T
AMERICAN COLLEGE TEST, 1961-1965
THE UNIVERSITY OF MISSISSIPPI

				67	ade Point	1				
Standard Score	Š	Below 1.00	1.00- 1.49	11	_ 	2 50 2 8	3.00- 3.49-	.50 .00	Mean	G Vi
	(2)	(3)	(4)	(5)	(9)	(7)	(8)	(6)	(10)	
6.1	41 G		₩ ₩ 67 /π	6 7 ((C)	<u>.</u> ආ	63 6		. എ ന	225
⊒ H 7 cq (2 4년	,r H	-1 GN i	0.2.0		၁ တ ်	ာ တ ။	-) (O) (I	i B
7.0	ა გ.გ.			1.0 7.		7 0 0 -	က ထ ၊	7.7	၁ က (1.03
224	လ ဝ ဗာ ဗာ			ਰਾ ਜੀ	о H	O∙o	70	ന. പ്രി	တ် ထ	
926	20			4.0	ထြ	ហ្វ	67 <		40	.52
\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	- 			Si of o	Ω	D 64	H 67	ff co	ညထ	3 (D)
0 0 0 0	٠ ۲					. H	% 4	77		9 / 0 /
	40						ရာ		ണ	60 r
72	N								41 ∕	o I
Motale	776	* T			101	60	56	.53		
2	н Н) H	1 1 2 2	+ - - -	3				All Control

TABLE IV.13

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTWITH COMPOSITE STANDARD SCORES OF GREATER ON THE AMERICAN COLLEGE TEST WHO ATTENDED THE 1961 SUMMER SESSION

	, V	(11)		
	Mean	(10)	2.22.23.33.25.25.25.25.25.25.25.25.25.25.25.25.25.	
	3.50-		H PERHA	27
ď	3,49 3,49		H40000H H4	19
E Average	2.50 2.99	(2)	1204011 H	15
de Point	00 49	9	podenia od d	14
Grade	1,50- 1,99	(5)	- പെ	6
	. 49	(4)	~ ~ ~ — ·	ည
	I W =1	(3)	H	cv.
		(2)	4 00 00 F P 00 C H 0 P H	8Ţ.
	Standard Score		- 1222222222222 - 12224222222222222222222222222222222222	Totals

Due to the small number of students enrolled during the summer of 16 the standard deviations and the correlation have not been determined

TABLE IV-14

FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER

THAN 18 ON THE AMERICAN COLLEGE TEST

THE UNIVERSITY OF MISSISSIPPI

END OF FIRST SEMESTER

	40 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
Mean*	——————————————————————————————————————	
3.50- 4.00		58
ရွိက က		104
nt Ave 2.50-	は	145
2.00- 2.49-		153
1.50-	ロヨココ コ ロタキア 4 5 4 4 7 4 8 1	102
1.00- 1.49	LACCHA ON H	87
Below 1.00	1478000 3178000 1478000	22
Proba-	U U C & C 4	52
Dis- missals	ONHOOONHH	22
With- drawals	o-l HαH	\Phi
No.	CD ∞ ∞ ω ₹ € € € € € € € € € € € € € € € € € €	709
J (11 C) L	1288888888888888 001888480280818	Totals

TABLE IV-15

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST THE UNIVERSITY OF MISSISSIPPIEND OF SECOND SEMESTER

	· · · · · · · · · · · · · · · · · · ·							**						•		(x_1, \dots, x_n)	
	S.D.	(14)	.70	. 85	.77	.81	80	.64	· 80·	.78	.62	. 49	.67	. 47	က က		•
	Mean	\sim	())	2.10	. c.	S	ണ്ട	4	ຜູ	-	တ	4	מ	ស	!		٠
	3.50- 4.00	(12)	~	∾ ⊢	S	87	Ŋ.	က	ဖ	7	ילא	~	4	ഗ	-	55	
	rage 3.00- 3.49	(11)	7	ထ တ	14	14	10		Ŋ	Φ	41	ب م	7	N	ญ	91	~
	t Ave 2.50- 2.99		20	ဘ က	16	15	14	14		디	ω	(י כיי		•	143	
	2.49	(6)	C e	70 70 70	30	က (၂	97	전1	10	ഹ -	က ၊		-1			191	
	Grad 1.50- 1.99	(8)	25,	급하	හ උ .	75 15		- 	တ ု	∾ (ď	7	-	•	92	
	1.00- 1.49	(2)	17	× & Z		ထ (ယ (⊢ (ا روز				•		4164.	72	
	Below 1.00	(9)	დ (က က	4!	ഹ (က ,	,	-1 (~						37	
,	Proba- tions	(5)	410	0 4		က (ء 0	r	⊣ (N		•			-1	32	
	Dis- missals	(4)	~ t	, 4!	4 1 ¢	ا ترا	1	•			•		•			26	
	als	(3)	H o	ગં ~- ા		7 -	- -	- (- (- (- (- (- (- (- (- (- (- (- (- (-	-1	•	-1				2 ,	11	
	No.	[2]	9 6 8 8	28.0	ω F ω Γ	٦ ر	4 E	٦ -		# ፍ ኃ ፍ) -) (X	> <	H "	662	
Stan.	dard Score		5 C		70	26	3 C) ()	500	79	00		or.	\ \ \		Totals	

TABLE IV-16

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN
WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18
ON THE AMERICAN COLLEGE TEST WHO
ATTENDED THE 1962 SUMMER SESSION
THE UNIVERSITY OF MISSISSIPPI

TABLE IV-17

C AVERAGES AND DISPOSITION OF 1961-62 ENTERING WITH COMPCSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST THE UNIVERSITY OF MISSISSIPPI END OF THIRD SEMESTER GRADE POINT FRESHMEN

Stan-				•			Grac		t Aye	rage		,	Î
dard Score	No.	With- drawals	Dis- missals	Proba- tions	Below . 1.00	1.00- 1.49	1.50- 1.99	2.00- 2.49	2.50-	က က	3.50-	Mean	, C
0	(2)	(3)	(4)	(5)	(9)		(8)	 	10)			(13)	(14)
တ နာ	70	S	87	റാ	4	1	74	91	13	7		C	99
02°	က (၁)	න ්	673	9	ဖ	₹#	Z	12	တ	. ပ		2.05	.75
	် (၁)	₹ # (- -	رى	9	Φ	겁	10	15	ထ	40		· Q
776	4, 6	7	י כיו	4	ဖ	œ	21	12			N	-	.80
, c	n (G	•	က (7	ဖ	12	10	တ	<u>၂</u>	m	8	. 79
4 4	か。 サ -	77 -	4 ; ((C)	ဖ	ro.	4	H	တ	တ	ဖ	4	1.02
0 4	ֆ. e Հ	41 G	72.0	বা (1 20	ના ∙	S)	ထ	ω	10	6 2	4	98.
200	ن ن ن	7 6	, M	m (4 1 (41	ဖ	7	87	∞	~	-	98
> 0	- つ つ で	7 r		79 n	N	٠.	ന (က	_	_	7	Φ.	88°
00	ر 2 د	- 1		F		r	က	41,	က	4ા	4	9	• 78
) C	1 r	_		-		1 r			~	₩.	က	.	.74
) (r	ન ન	⊣						2		က	က	တ္	.70
1 (4 2, 6) 4		6	*		Í					41	9	.26
Ö Vi	41					_	•			8	႕	9	86*
Totals	525	26	21	35	α •:	5.4	16	44	60	α	OG		
4	:		•)) .)	1	H):	5	1	3	0		
**													

TABLE IV-18

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST THE UNIVERSITY OF MISSISSIPPIEND OF FOURTH SEMESTER

-50.							E	1	Γ	rage			
gard Score	No.	With- drawals	Dis- missals	Proba- tions	Below 1.00	1.00- 1.49	1.50- 1.99	မော တ	P: 0	30- 3.00- 39- 3.49	3.50-	Mean	c,
(7)		(3)	(4)	(5)	(9)	(2)			日	(11)	·1	(13)	
O) (65	~	-	2	·	ဖ	13	18	15	œ	2	6	80
)) ()	44 d	6	c4 -	r)	6 7 1	9	ထ	13	ത	ග	~	2.27	.78
18) () ()	4	41 C	ب ب	හා (II'	ထ	12		œ	87	<u> </u>	.78
3 6) G		7 9 G	, N r	N 1	φ (12	67	72	H	ഹ	2	.75
) 7 3 3) •	*	ý 6	- 1 ċ	- 1 (တ (毋,	14		12	ಉ	7	.81
4 C	e P	1	N) G	7	က (~	က	10	12	ဖွ	7	לא	0
3 G	\ C D 6	N :	N (N (*	4	12	ဖ	디	~	ຜ	.77.
5.0	9 6		7 r	- +	77	-	ıO.	4	Ŋ	ß	7	9	96
- 00 - 00) d	•	- 1		-	•	Ń	_	.	2	S	6	.75
) () ()) 	•				~	~	~	ഗ	က	ဖ	0	.81
) C	4 C							~	H	Ŋ	₹		. 54
) [) Մ		•	•				 1	~	ഗ	က	2	43
	ۍ د		•		•				•	က	~	4	.43
	5		-1		⊣ ·	*				~	, H	31	1.66
TOTATS	4.78		10 1	16	18	42	19	116	93	94	20		•
					•			•)		

TABLE IV-19

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRE WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18
ON THE AMERICAN COLLEGE TEST WHO
ATTENDED THE 1963 SUMMER SESSION
THE UNIVERSITY OF MISSISSIPPI

Sean							- 1		ļ				
Score	No	With- drawals	Dis- missals	Proba- tions	Below 1.00	1.00-	1.50 1.99	2 49	200	rage 3.00- 3.49		Mean	S.D.
		70	12)	707	(0)	S	(8)	(8)	(10)	(11)	(12)	(13)	
-1 e 2) C	-1 FG		•	r		ଷ	~	87	Ф	83		ÇTŞI W	64
	- 07 		-1 ←	-	.7		∾ (ဖ	(C)	~	~	2.44	.68
186	2 6				:	•	י נים	ကျ	4	ፍጋ		4	54
1	1 1		1	•	v.		 - (_	 1	7	41	-	.74
4	1 cc		•	- 1		-4	77		(24)	N)	ぜ	O)	.85
ļ Ķ									, 		ယ	'n	\$ 50
) (G)) r- H					•	~	Ŋ	က	က	~	φ.	* 70
22	16	•	•					က	I	ঝ	01	0,	.64
200	· 67					•	-1	• .	~ -1	κ'n		ᅼ	က္မ
8)),			4		-			•	7		Ω.	ليس
	N				•				-		;	Ç	: •
	 (•		4	-	- 1	A C	24.5
3	N		• • • • • • • • • • • • • • • • • • •	•	•		• **		, ,	- -) () ())) (
						21.			1	4		•	02
Totals 127	127		ĠΥ	Ģ		ដ	9						
		1	•	3		D	77	97	7 6	35	5 2		
			2										

TABLE IV-20

GRADE POÍNT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST THE UNIVERSITY OF MISSISSIPPI END OF FIFTH SEMESTER

	-		, .		.	•	H	0	int Av	(1)			
derd Score	No.	With-drawals	Dis- missals	Proba- tions	Below 1.00	1.00- 1.49	1.50 1.99		2.50- 2.99	3.00	3.50-	Mean	S. J.
	(2)	(3)	- (4)	(5)	(9)		8	6				(13)	-
ø	47		i.	Q	•		£	G		(,	9	1
) (C	\ G		ન	Ó	⊣ ,) T	<u>, </u>	٦ ا	9	מכ	- -i	7	.74
	7 (n) (n	1	- 4	~	Ŋ	14		10	တ	ů	.70
7	7.4	4	%	~	, 	ന	ശ	10		17	87	, S	•64
7	79	-1	N	6	~	ണ	_	17		13	ဖ	S P	တ္
77) 180	က က		.	8	1) -	C3	တ	ဖ		11	က	ယ္	.65
3	4 0		· ·	0		ഹ	—	က	15	2	တ	2.84	60
	20			l	٠.,	H	7	∞	<u>ထ</u>	-*, Φ	ഗ	9	.70
6/ E	# (7)	k		-1	, i	1	න	Ś	ਚ		7	7.	.71
7	00			į.	. ` `		•	₹,	7	10	വ	0	.51
9 6	o c		24 4 %			 1	~	~	ঝ	S.	4	\Phi	.75
	D o	Agr		1 2 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			•	∾.		∾.	41	4	. 62
	0 4		•	- 1				•	~		4	4	88
7 6	}		i j			•			~		က	4.	53
	מי					• • • • • • • • • • • • • • • • • • • •		•	<u>-</u> -1	- 1	–	3	. 61
			がある。		2				,	z			٠.
- a	167		ing and the contract of the co	Ç	\` -			•					
	+ 5*		•	<u>`</u>	41	27	43.	86.	111	တ်	54		

TABLE IV-21

"tv . I					,	•					• • • • • • • • • • • • • • • • • • • •		
	G.D	1	99	ලී සූ	79	72	88.	*57	.60	, w. r.	00.		
	Mean	(23)	ا د ملک	ap (p	7.5		, W	<u>ල</u> අ	. 60 c) () () () () () () () () () () () () ()	-1		!}:
. 1	3.50-4.00	(12)	io.	ત્રા બા	ဟ တ	~ v:	(it) it	ય) W r	-	65	
	3.49	(11)	ω	12 C	10	01 8	(C)	ન છ 4	- ده ا		7	97	
	2. 50-	(10)	ю. Н	101 111	8 11 12 13	119	-दा	رم من ا	— с	3 (2)		105	
	2.49	(6)	H.	- - - - - - - - - - - - - - - - - - -		126	S	8		_		83	
13	0 8	8	~ 6	्र ५ ४ ।	ဝ က	സ ⊢1	₩,					37	
	1.00-	3	က		% ⊢ 1	e e	~ ~					검	
	Below 1.00	(0)	♂		કન્ન _.	- 	, in		ar que			01	· }
	roba-	72	ر ا	\$) <u>} </u>	8.7	ନ୍ଧ ₍₋) 	L second	6.00	() -	63	
.3.	13. T. P.					(N)				7		r-1	
	Dis- missa	1	4-		i A.	-11 -			(1) (1) (1) (2) (1) (8)			N	
	th- 2 3 3					. Tek							
	0 dr									rii),	-0		
The state of the s	No (2		ශ් නි		(a) (a) (b) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c		3 63	9		(a) (b)		Ls 405	
8			28	a a	22			25	18				

ಹಿ

43

20

87.

 ∞

TABLE IV-22

	1	w .			Ú.
			•		
	2	4		,	·· ·
	1	ł			
•	2	•			
	7	•		. •	
		ļ			
	5				
	1	í			
		00			
	2	,	ĺ		
		بواا	•		
,	2				
	8				•
1.		E			
	Z				
	9			-	
`	~		X	F	14
	6		世	Z	Σ
	$\tilde{}$			10	E
-5,	-	2	H	ξĎ	H
	196	C	S	H	Ŋ
•	O		띰	O	Ŋ
		×		ند	H
		·	5.7		XX
	6	CO	छ	昱	14
		Ø	5-1	呈	Z
	X	SE		5	_
	9	5	펏	W	H
		X	×	-40	ب
		C.		3	5
	מ		Z	Ŏ.	H
	Ø	α	×		H
	P	K	Ŋ	. .	Ø
	\mathbf{g}		N	田	K
	Ξ			出	본
	-		불	Ç-1	
		to	7		z
	إسبا	114	•	r-i	11-
. 1	H		100		لببنغ
	ANI	H	田	百	سن
ا میرم	S AM	E	国		国
1	ES AND	SITE	間	END	田田(
1	GES AND	SITE	N THE	THEN	国国
1	AGES AND	POSITE	ON THE	ATTENDI	
1	RAGES AND	MPOSITE	ON THE	ATTENDI	THE
1	TERAGES AND	COMPOSITE	ON THE	ATTENDI	
1	AVERAGES AND	COMPOSITE	ON THE	ATTEND	四田
1	AVERAGES AND	H COMPOSITE	S THE	ATTENDI	
1	T AVERAGES AND	TH COMPOSITIE	到 No	ALTENDI	
1	NT AVERAGES AND	TITH COMPOSITIE	到 I	ATTENDI	
***************************************	DINT AVERAGES AND	WITH COMPOSITE	国 国 国 国	ATTEND	
	POINT AVERAGES AND	WITH COMPOSITE	田田の	ATTENDI	
	POINT AVERAGES AND	WITH COMPOSITE	ON THE	ATTEND	
	E POINT AVERAGES AND	WITH COMPOSITIE	ON THE	AUTEND	
	DE POINT AVERAGES AND	WITH COMPOSITE	ON THE	ATTEND	
	ADE POINT AVERAGES AND	WITH COMPOSITE	ON THE	ALTEND	
	TRADE POINT AVERAGES AND	WITH COMPOSITE	ON THE		
	GRADE POINT AVERAGES AND	WITH COMPOSITE	ON THE		
	GRADE POINT AVERAGES AND	WITH COMPOSITIE	THE NO		

S. D.	© 00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Kean (13)	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8
3.50- 4.00	31443244321 2
3.49 (11)	ららよららさような1121
2.50- 2.99	4 4721 2
2.00- 2.49 (9)	480004 ይገ '8 ሀ
Grad 1.50- 1.99 (8)	NH NHHH
1.00-	0 H0 HH
Below 1.00 (6)	H H
Proba- tions (5)	
Dis- missals (4)	
With- drawals (3)	
de rd Score No	18.28.28.28.28.28.28.28.28.28.28.28.28.28

TABLE IV-23

GRADE POTA AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESTAND WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST THE UNIVERSITY OF MISSISSIPPI END OF SEVENTH SEMESTER

		A STATE OF THE STA		4				Ź					
	1						E	٩	t AVB	rage			
Score	No.	//th-	Dis- missals	Proba- tions	Below 1.00	1.00- 1.49	1.50-1 99	2.00 . 2.49	2,50- 2,99	3,00- 3,49	3,50-4,00	Mean	o. D.
3		(3)	(4)	(2)	(9)	(4)	L	9	2		-	(13)	(14)
HAY WANAWANDER OCHNEROLES	後ののまでの女士女子のこれですることできるなりの143742	8 . H H	нн м нонн	мним н		4·12/2/10/10/10/10/10/10/10/10/10/10/10/10/10/	യയയെയെപ് പപ പ	ひままままままらまる	uu oo4roooos	L T TH THE TERMS TO THE TERMS T	るきまらまできまざらまま	88888888888888888888888888888888888888	1
<u> Potals</u>	325	41	10	တ	ထ	다 [හ භ	64	77	80	45	. ,	

TABLE IV-24

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST THE UNIVERSITY OF MISSISSIPPIEND OF EIGHTH SEMESTER

1	S.D. (14)	7.000 C C C C C C C C C C C C C C C C C C	
	Mean (13)	323332332323232333 32333773333 323337733374 323337733374	
	3.50- 4.00 (12)	469696994446944	42
	ge 3.00- 3.49 (11)	<u>н</u> 440000000004411	75
	2.50-2.99 (10)	こ らりりりり しょしょ	64
-	2.00-2.49 (9)	∞ ω, 4 ω 4 м м м м м м м м м м м м м м м м м	39
1	Grade 1.50- 1.99 (8)	തമ്പെയ് പപ പ	18
	1.00-	МНМН М	œ
	Below 1.00 (6)	rd rd rd	4
	Proba- tions (5)		41
	Dis- missals (4)		
· · · · · · · · · · · · · · · · · · ·	With- drawals (3)		
	No.	8 2 2 2 2 2 2 1 1 1 1 2 2 2 2 2 2 2 2 2	251
	Score Score	28222428888888	Totals

TABLE IV-25

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18
ON THE AMERICAN COLLEGE TEST WHO
ATTENDED THE 1965 SUMMER SESSION

	S.D.	1 14.0 20.0 20.0 20.0 20.0 20.0 20.0 20.0 2	\$
and the second		(13) 2.47 2.37 2.45 2.65 4.00 4.00	
	3.50- 4.00	12 8 8 8 12 13 13 13 13 13 13 13 13 13 13 13 13 13	6-
	3.00- 3.49	E 2440044 24	17
	Average 2.50-3 2.99 3	(OE 20 20 20 20 20 20 20 20 20 20 20 20 20	r.
	2.00-2.49	(a) www-44	15
1	5rade 1.50 1.99	(8) HH HMHHH	∞
	1,00- 1.49	8	87
	Below 1.00	(6)	က
	Pro- bations		~ i
	Dis- missals	4 2 H H H	Ŋ
	With- drawals		
	No.	S Courte o o al	73
LEST S	dard	1898888888888888	Totals

III. SUMMARY OF REPORTS

The preceding tables of Chapter IV show the enrollments in schools, by semester, and the grade point averages and disposition of the entering freshmen of 1961-62 who earned scores of greater than 18 on the American College Test. The enrollments indicate a drop from an initial enrollment of 709 during the first semester of 1961-62 to 250 at the end of the eighth semester. However, 67 students had graduated at the end of previous semesters or summer sessions. There was greater correlation between the scores on the American College Test and grade point averages for the group of students which had scores on the American College Test of greater than 18 than for either the group with scores of less than 15 or the group with scores of 15, 16, 17, 18. The correlation coefficients for the above group, ranged from a high of .381 during the 1962 summer session and .378 and .364 during the second and first semesters of enrollment, respectively, to a low of .140 during the 1965 summer session.

The standard deviations of scores, in general, were less than 1.00 for all scores greater than 18 which was the case for the groups discussed in Chapters II and III. An exception among the high group occurred with the four students having the highest score of 32. One of these withdrew at the end of the third year after receiving probation one semester and dismissal during another, but after being readmitted he earned a 4.00 grade-point average during his final semester of enrollment.

In Table IV-26 the enrollments have been summarized by schools on the basis of scores on the American College Test.

Table IV-27 presents a summary of the grade point averages and disposition, by semesters and summer sessions, of the 1961-62 entering freshmen with composite standard scores of greater than 18 on the American College Test.

In Table IV-28 a complete summary of the grade point averages and disposition, by semesters and summer sessions, for individual scores of the 1961-62 entering freshmen with composite standard scores of greater than 18 on the American College Test, is presented.

TABLE IV-26

ENROLLMENT OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST,

THE UNIVERSITY OF MISSISSIPPI 1961-62---1964-65

	t		Yea	r of E	nrollm	ent		
School or College	1st	1-62 2nd Sem.	196 1st Sem.	2-63 2nd Sem.	196 1st Sem.	3-64 2nd Sem.	1964 1st Sem.	2nd Sem.
(1)	(2)	(3)	. (4)	(5)	(6)	(7)	(8)	(9)
Liberal Arts	428	386	286	247	203	192	138	107
Business and Government	123	136	119	120	119	114	: 102	87
Education	49	55	55	58	54	55	43	25
Engineering	92	68	48	37	27	27	24	17
Pharmacy	17	17	17	16 '	18	19	. 18	15
Totals	709	662	525	478	421	407	325	251

TABLE IV-27

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18
ON THE AMERICAN COLLEGE TEST, THE UNIVERSITY OF MISSISSIPPI 1961-62---1964-65

GPR and	1961		1961-62	2		1962-63	က	*	963-64			964-65	
Disposition by Scores	S	1st Sem		ki ki	1st Sem	2nd Sem	8.8	lst Sem-	2nd Sem-	V.	1st Sem.	2nd Sem	v.
6		(3)	1 1		(9)	(2)		(6)	100		(12)	(13)	(14) (14)
Enrollments	1 8	709	662	120	525	478	127	421	407	143	325	251	73
Withdrawals	A.	∞	П	٠.	26	4	r-l	 1		₩.	, VI	H	-
Dismissals		2.2	36	, - -	21	19	က	&	12	च ।	10	∞	S
Probations		52	32	4	35	91	~	20	133	• • • • • • • • • • • • • • • • • • •	တ	 ♣)
Grade Point Averages:					•	. •		•		•		;*	· .
00-1.4	വഗഠ		37	က တ ရ ရ	8 4 C	. 18. 42.	က	4.72	10	27.0	8 H 6	4100	നവ
2.50-2.49 2.50-2.99	니니 (4.70	1111 504 503 503	161. 143	70 70 70 70	100 000	116 93	7 9 9 P	96 1.1	387 25.	28°	64 77	2 8 8 2 9 4	
50-4.0	70	O 10	91 55	188	88 80 90	94 50	223	54 54	97	44 & & & &	45 45	422	144
Graduates	.	•		•		•		, 69	10	23	32	185	32

TABLE IV-28

GRADE POINT AVERAGES AND DISPOSITION OF 1961-62 ENTERING FRESHMEN WITH COMPOSITE STANDARD SCORES OF GREATER THAN 18 ON THE AMERICAN COLLEGE TEST, BY TEST SCORES, AND

			•						,							•	
5	SS	(15)	(73)	10			17		ဖ	Ó	4	N	٠.,		—	•	٠.
964-6	2nd Sem.	(14)	(251)	36	24	24	37	63	26	24	12	α	တ	က	ယ	41	8
	1st Sem.		(328)	43	31	ر ا	49	တ	2 8	5 3	8	21	14	വ	2	41	C/J
	S.S.		143)	21	10	18	27	16	∞	14	ထ	œ	in.	N	બ	ধা	
963-64	2nd Sem.		407)(54	38	48	63	33	ထ္ထ	ထ	24	22	16	တ	_	ۍ. س	က
	lst Sem.		(421)(22	42	49	64	99	40	37	24	5 6	18	ා	ω	က	က
က		(C)	(127)	13								2	က		~ 1	- -	7 7
2-6	2nd Sem.	(8)	(478) (1	65	49	09	69	48	4 3	ლ 2	53	30	ත ෆ ්	너		က	က
	lst Sem.	(3)	525) (4	70	55	65	74	5 3	49	42	35	al S	20 100	=		J.	4
	S	(9	0) (0	₩.	တ	- H	හ	- - H	~~ I	ന	no.	لحد	က၊		∾ 1		
· []	nd em. S		709) (662) (120	9	 		 1	<u></u>			•			. ,			e i
FI	н 82	4) (9) (6	<u>ගි</u>	7	7	တ	^	ယ္	(J)	4	ണ	7		-		
			(70	10	œ́.	∞ \	တ	_	ထ်	<u>ഹ</u>	4	က	2	- 1	H)	•	
196	, M	(3	(81	14		0		_	\	x	ഗ	 (က		> •	-	
	nrol ent	(2)		108	자 () ()	27 827	တ္ (တ)	75	64	56	<u>4</u> ان	M	27.	ე (7	X	4
PA and	DISPOSITION DV Scores		Enrollments:	O (7		7	2	7	22	917		9 G	7			N

137

IABLE IN-28 (continued)

			137
S. S.	(1)	H) M H H H
964-65 2nd Sem. (14)		-	किन्न निर्नन
1st Sem.	(4)	ом. н н	ннанюнн
S.S.	(2)	3	н нн
2nd Sem.			
lst Sem			
63 S.S.	Į.		ннн
1962-63 r. 2nd r. Sem.		2 2 3	1242222
lst Sem	(26)	0040 0400H H	
961-62 2nd Sem S.S (5) (6)			
4 e st	(8) (11	он номи понномини о понномини о о понномини о <	
961 1 S.S. S (3) (он нон. 2	
Initial Enroll— ment (2)		Д Ф Ф Ф Ф Ф В В В В В В В В В В В В В В	01 08 80 40 07 40 07 40 07
GPA and Disposition by Scores	Withdrawals:	E 32 22 22 25 25 25 25 25 25 25 25 25 25 25	

TABLE IV-28 (continued)

1963-64 1964-65 Ist 2nd Sem	
1963-64 Znd Ist Sem S S Sem (11) (12) (13) 5 3 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
1963-64 2nd Sem S (13) (13) 1	
1963-64 Zha Sem (11) 13 3 3 11	
1 1 1 miles 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
S H H G G G G G G G G G G G G G G G G G	ŧ
252-63 252-63 3 Sem 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	
(3) S S S S S S S S S S S S S S S S S S S	
(4) (2) (3) (4) (6) (6) (7) (7) (7) (7) (7) (7) (7) (7) (7) (7	
1961-62 Sem (5) (32) (32)	H
い。 11 13 13 13 13 4 4 13 2 4 4 2 2 4 4 4 4 4 4 4 4 4 4 4 4 4	
[38] S S S S S S S S S S S S S S S S S S S	
Enroll- ment 122 22 22 22 22 22 22 22 22 22 22 22 22	4
25 23 23 23 23 23 23 23 23 23 23 23 23 23	
#18 position 8 20 20 20 20 20 20 20 20 20 20 20 20 20	200

TABLE IV-28 (continued)

	# 3				
	S.S. (15)	(5)	•	8	H
64-65	Sem.	3	न नन		87.88
61	Sem. (13)	(8)		2	414004
	S.S. (12)	(2)		E	∞ ⊣ ∞
963-64	Sem. (II)	(az)	स ं ललनन	3	စာ စာ ⊢
	Sem. (10)	3	HHHH	(27)	_ _ _ _ _ _ _ _ _
	S.S.			9	0 4
962-63	Sem.	(18)		1 (42)	ၜၜႜ႞ၜၜ
14	Sem.	(38)	40000N0N4N	(54)	႕ 4 ထ ထ ထ
	S.S.	<u> </u>	H 04	6	4 – –
29 <u>-196</u>	Sem.	(37)	യ ാ പെ പ രാ പെ പ	7 (23)	H H H & &
	Seit.	25	447000H8H	(8)	レキミッエ
1961	3 (3)	. 2		9	м н н
Intital finedia	ment (2)		0 8 8 9 F 0 R 4 8 Z 1 -	4 00 44	ಹ ಈ ಕು ಎ ಗು
	Scores (1)	Point ges:	224384466 <u>588</u> 6	9 H W H	တဓ္မလ္က
		Grade Pos Everages Below 1.			HUNDN
	T-10		一种 通过多点的 海南 的 医二氏素管纤维性纤维性炎 医二角	The second of the	

TABLE IV-28 (continued)

1		
V olo		Ф нн нмннн
00		
No sta		
200 St. 1		о миммим на на Н
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	~~	
	The second of th	
13.5		S NH NHHH
60		
963-64 2nd Sem. (11)	NAH.	ы
حساب السا		
Sem (10)	लिन नि न	は、 アちちアらユアき 。 な
- 10101		
الله ا		
8 8 O		N 10010 0 1
2nd Sem. (8)		
962- 2nd Sem (8)	7 H 7	ଳି
1st Sen		
1st Sen	odd eee	S HHLWL H 4VHLW4ROUU
8.8		и и ноои 4 инн
FO .		
2nd Sem (5)	ФН) Н	N DENEMOTORE &
Sem (4)	4,00H H	ひょうようようてよう エ
-II Vi-		5 6161 4 5 1111 1
9 S		ഉ നുപനപ പ
- dH		
SH 0 H	るられるなユー 4のなアならなの4	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Init Enro ment		
	the state of the s	
P. and		
6 # 8		1 00HNN400F000HN
and post Scor	NOW OF CHANDED	I AMMONNONNONO
i de di		

TABLE IV-28 (continued)

. \$		060				
10 mm		S.S.	(12)	889448	(15)	меме нау
	964-65	Sem.	(39)	∞ଦ4ଦ44ଥ [ା] ସ ୴ '		พ๛หญิงหอห
and the second s		ist Sem. (13)	(64)	り 工 4 4 4 8 8 8 8 8 8 8 8 8 8 8 8 8	(23)	oo4⊬ 00000
Jern T		5	(8)	40004 OH O	S	4 4 5 2 1
	64	N.) (2			
	1963-64	Sem.	(82	HUND4000 0	1 (105	
		Sem.	(88)	ユニココ 3.4076500004612	(111)	0 1111 0 7 0 1 3 0 4
		S (S	26)	പരഖം വയ	26)	രസ4പപ്പെ
	-63	्व व)			
	<u>o</u> l	Sem (8)	911)	ロエエユエコ 8 8 6 0 4 0 8 4 7 1 8 1	(93	1 11 1 0 1 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8
		Sem (7)	(61)	ままれまれるアラダーの	(92)	Ц ЦЦ вор40000
	وديار	S.S.	(29)	ച ക്യരയെയ്ല്പ് പ	- (E	888888
	<u> 9-196</u>	Znd Sem.	61)	8 4 8 8 4 4 4 8 8 8 8 8 8 8 8 8 8 8 8 8	143)	84444 60000440
	Į.	Ist Sem.	53)(1	20040200000000000000000000000000000000	45)(1	8112811 84401888
	~1	77	0			
	36 T	n in	(14	vetant de d	(15	4W04W44
	. T					
	Initi	Enrol ment (2)		Ц С 8 8 0 К 0 М 4 8 2 Ц Ц 8 4 2 0 К 4 9 0 К 2 8 2 8		01 00 00 00 00 00 00 00 00 00 00 00 00 0
	D.		Ó			
A CONTRACTOR	and	ositi ores	3	ᲗᲓ ୷ ଊଊ୕୴ଊଡ଼୵ଊ୕ଡ଼୕୕ଌ୕୷	0 0 N N	മെപ്രത്യെ
	THE PERSON		8	ัก จัง หัก คือ	50-	-
A. 16.	š					

					e i
	S S (15		5	NHOOHH NH	
	964=65 2nd Sem. (14)	തപപപര	32	44000000000	
	1st Sem. (13)	က် ဖျ	(80)	Head Hanned	::-::: 73 - { : .:
	S.S. (12)	~	(43)	らら4002432112.	-
	963-64 2nd Sem. (11)	സയ പമമ	(97)	00000000040	-1 1
	19 1st Sem. (10)	てもこととこ	(95)	のひはむはとのとひらなる	-
c Inued)	ဖြ		35)	<i>ผ</i> าพ <i>ะ</i> กนผ _่ นผ	-1 r1
	962-63 2nd Sem. S (8) (യഗപപ	94) (3	യയയപ്പെന്ന ലസ സംപ്രപ്രമാരം	o —
	19 1st Sem. (7)	P 80 4 H	88) (8	გი დ ი ი ი ი ი ი ი ი ი ი ი ი - ი ი ი ი - ი	4 PV
	(2)			ସସ୍ପ୍ରମ୍ୟ ସ୍ଥ ମ	
	Snd (5)	H. W. 100 H	91) (18	20044000004000	1 (2)
	1961 1st 2n Sem. Se (4) (5	တ က က	4	NOOUPFOOONNO HHH	1 ml
			3) (10	$-\mathbf{N} \otimes \mathbf{N} \otimes \mathbf{N} \otimes \mathbf{N} = -\mathbf{I} - \mathbf{I}$	
	3				
	Initia Enroll ment (2)	ಬ್ಬಳಕ್ಕ ಬ್ಬಳಕ್ಕ		7 % & K & K & K & K & K & K & K & K & K &	d
	구설 분 등				
	Sositi Cores			22HN93	*. * * * * * * * * * * * * * * * * * *
		No.		28487 A. V. 288	(10)

= TABLE IV-28 (continued)

(15) S.S.	(12)	N 0	พผษ			(32)	ಬ ಗು	• ୧୯ ୧୯
2nd Sem. (14)	(42)	4ા છા ન ા	မက္မေက	∞4H −	lw-i-i		0 9 L 8	1 1 2 4 4 4
1st Sem. (13)	(45)	N 60 N	Q 41 L W	w w w H	I പെപ്		v4 ω Φ 4	10H0
S.S. (12)	(33)	м н 4	ન જ જ ના	4°00	7	(23)	თ ქი	୷୷ୠ
1963-64 Znd Sem (11)	(65)	N 41 41	က္ တင္ကာ လို	დ ν ν 4	оо п.	(10)	н 0	н юм
Sem. (10)	(54)	പ്രമ	ပ က္ကေတ ဟ	ৰুত কুৰ	4004	8	^	
S. S. (9)	22)	N	4400	ભ ત્ન				
962-63 2nd Sem. (8)	(20)	M M M I	0 W F Ø I	႒ဖတ္	ଟ ବ୍ୟ ଲ			
Ist Sem.	(39)	က	M -160	なて48	ი 4 ⊢			
(6)	(18)	МHг	୳ଊଊଊୢ	, הרק	' 1			
961-62 2nd Sem. (5)	(52)	พพาแ) (A (1) (A)	0 L W L	4 ₁ 0 H			
Ist Sem. (4)	(58)	너 너 야	ာ လ လ လ င	0,000	ာ တ			
38.5 38.5 38.5 38.5 38.5 38.5 38.5 38.5	2	Н. 6	J പ യ യ പ	1 77	•			
7.								
Initi Enrol ment (2)			7.7.8 7.0.4 1.0.4	E N E L	, , , , ,		9000 P	
nd sit(on ores	8							
Fano Sposi Scor	50-4.	22 d %	80888 8480	N 00 00 00	31 32 37 37 37 37	် ရ	2H88	00 00 44 00
Baa	m				, U	D S		

TABLE IV-28 (continued)

2. E.	ر د اندران		1		
1			•	\neg	
5 y			ល	2	HAH
	, (1) (1)		rol		
70° 2	65	49.			
		O	E	4	全てのころよる。
	64	2n	r)		AHA.
	96		"		
		لبد	ė	8	HORON
	.7-	W	W		
		-1	U		
	H	. d 50. j	- 20 - 20		
,X∳° 'Y		, D.	S	2	ന ന H
	1.		ť		
	4		נט		
	9-		m.		
	က	nd	e _I		
***	96	2n	വ		
, v					
n	3 6 e 2 9 -	st	He	0.1	
v.		H			
۸.		e G			
7					
te e	γ.I		S.	6)	
			S	_	
8	63		1	1.	
	-7	nd	m	()	
	396	22	Se	8)	
	တ				
,		L¥	Ħ		
	•	1.5	W	7	
		ia-r	U		
5,4	\mathcal{F}_{ij}		*		
• it			t)		
	E.	48 -	מ	(9)	
3	7				
	9	70	ធ		
	19	Ĕ	0	2	
, T	36	"	LO		The site of the site of the second of the site of the
	Η	4	u.		
		i w	Ø	(7	
	V.	<u> </u>		-	
				2	
() 3		ľ	to.		
	16			က	
4	ر ب		t/a		
	ď.				
4	Ä	1	4.5		two the way of the
ا ا	Q	口		_	
	111	0	4	C)	ちておるるの生
	-04	nr	en		400HH
1		ন্ত্ৰ	Ĭ	S (V	
		ε,	11	1.00	
		Ħ	g. 2	رم. بر	
d.		-H	Ø	47	
±. de		44	O		
, <u>1</u>	E	O.	Q		
	Ю	Q.	S CO		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
	1	U)			
1		ä	d	, ,	
€.		*****	4.	5	100 本 100 100 100 100 100 100 100 100 10

Chapter V

HOLDING POWER OF THE UNIVERSITY OF MISSISSIPPI BASED ON COMPOSITE STANDARD SCORES OF THE AMERICAN COLLEGE TEST AND ACADEMIC ACHIEVEMENT OF 1961-62 ENTERING FRESHMEN

How many students continue at The University of Mississippi for four full academic years? How many students graduate prior to the traditional four year enrollment period? How many in the initial enrollment leave after the first year or even after the first semester? Does the academic ability indicated by the standard score on the American College Test have any effect on the dropout rate at the University? What is the academic achievement of students based on their college entrance test scores? The following tables indicate answers to these questions.

Table V-1 indicates the semesters of attendance at the University for all the 1961-62 entering freshmen based on composite standard scores on the American College Test.

Table V-2 shows the graduation and disposition of the 1961-62 entering freshmen. This includes a complete summary of the number of graduates, the number of students enrolled after the 1965 summer session, the number of dropouts throughout the period of enrollment 1961-1965, and the number of transfers as indicated by the transcripts which have been issued. A supplement to Table V-2 indicates the number of graduates for each semester and summer session of enrollment through the 1965 summer session.

Tables V-3-A, V-3-B, V-3-C, and V-3-D summarize the mean scores, the standard deviations, and the correlations for grade

point averages by years of enrollment of the 1961-62 entering freshmen based on composite standard scores on the American College Test.

957 196

Totals

TABLE V-1

SEMESTERS OF ATTENDANCE AT THE UNIVERSITY OF MISSISSIPPI FOR 1961-62 ENTERING FRESHMEN BASED ON COMPOSITE STANDARD SCORES OF THE AMERICAN COLLEGE TEST

i	
S.S. (14)	2425600000000000000000000000000000000000
964-65 2nd Sem. (13)	u 4441717864478645868648
1st Sem. (12)	1122246846844444444444444444444444444444
S.S. (11)	u mmungmanda n mmungmandnomrom4 n mmungmandnomrom4
963-64 2nd Sem. (10)	12224458468882241 814564816888888884569758
lst Sem.	188845446848831 487644878078945074689858
S.S. (8)	さる4487023731485173 よこぶ
2nd Sem. (7)	
lst Sem (6)	uu
S.S. (5)	
961-62 2nd Sem. (4)	111240000720048211 500000000000000000000000000000000000
Lst Sem.	1 121212000 C D S 8 8 9 C 6 2 4 8 2 1 1 1 2 8 6 C C C C C C C C C C C C C C C C C C
1961 S.S.	പ പംബ4 ऴറ ൻऴ4ൻറെല്ഗ്യൻപെ ഗ്പ
Standard Score (1)	# 0

TABLE V-2

GRADUATION AND DISPOSITION OF 1961-62 ENTERING FRESHMEN BASED ON COMPOSITE STANDARD SCORES ON THE AMERICAN COLLEGE TEST THE UNIVERSITY OF MISSISSIPPI

Grad-	Enrolled				
uates	Spring or	-	No. of	, a	
thru	S.S. 1965	No. of	Students	•	
Standard S.S.	but not	Students	Trans-		
Score 1965	Graduated	Dropped	ferred*		Total
M F	MF	M F	MF	M	\mathbf{F} \mathbf{T}
(1) (2) (3)	(4) (5)	(6) (7)	(8) (9)	(10)	(11) (12)
(D-1)					
(Below)		ó 7		•	
(10) 1 10 1	2	3 1	4	9	2 11
	4 1	3 2 1 5	Z 7	5 11	3 8 7 18
12 1	8 # T	1 O	5 1 4 1	7.T	7 18 6 20
13 1 1	ง จ	· 3 0	9 9 # T	14	6 20 8 17
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	4 1 6 3 6	3 1 3 2 1 5 3 5 4 2 - 2 - 8	4 2 5 1 4 1 2 3 _63_	- <u>1</u> 8 - <u>3</u> 1	$ \begin{array}{r} 8 & 17 \\ -\frac{11}{20} - \frac{29}{51} - \end{array} $
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	7 1-	8 8	· 136- ·	- 3 7- ·	$-\frac{11}{20}$ $-\frac{29}{51}$
16 5 8		6. 6	16 8	38	25 63
16 5 8 17 9 7	$\overline{12}$ $\overline{1}$	4 2	11 17	36	27 63
18 11 8	18 3	412	12 10	40	<u>3</u> 3 <u>7</u> 3
<u> </u>		9_ T 9_ _	$\frac{12}{17}$ $\frac{10}{13}$ \cdot	$-\frac{40}{59}$	$-\frac{33}{49}$ $-\frac{73}{108}$
20 14 11	8 1	16 9	15 10	5 3	31 84
21 21 11	$egin{array}{cccccccccccccccccccccccccccccccccccc$	6 6 6 7 5 10	15 13	5 0	32 82
22 31 11	14 1	6.7	16 13	67	32 99
23 10 13	8 1	5 10	18 10	41	34 75
24 13 8		9 8	10 4	43	21 64
25 19 14 26 11 8	3 2 3 1	2	. 18 6 5	42	14 56
26 11 8 27 13 8 28 13 1	4	11 2	6 5	30	15 45
28 13 1	2	<u>.</u>	7 4	23	14 37
29 6 1	7 4 7 7]. 9	4]	19	3 22
30 5 1	ำ 1	2 1	2 1 2 3	9	4 13 2 12
31 5	с.		3	10	2 12 8
29 6 1 30 5 1 31 5 32 2		. 1.	ĭ	4	4
3	, 4 Y	••••	•		
Totals217 136	190 70	204 116	900 355	000	200 7000
Totals217 136	139 18	104 116	209 123	669	393 1062

^{*}The assumption is made that all students who requested transcripts to be sent to another institution transferred to it.

TABLE V-2 (supplement)
TIME OF GRADUATION

				* · ·				
Standard Score	5th Sem.	6th Sem.	S.S. 1964	7th Sem.	8th	S.S.	,	M - 4 - 7
	MF	MF	MF	M F	Sem. M F	1965 M F	M	Total F T
(1)	(2)(3)	(4)(5)	(6)(7)	(8)(9)	(10)(11)	(12)(13)	(14)	(15)(16)
(D-1)	 	***						
(Below) (10)				, ,	-			
10	nt k		' • ·		1	•		$\frac{1}{2}$
11	•	· ·		· · · · · · · · · · · · · · · · · · ·	en e		ī	7
12		-	1		a ,	ī	î	1
13	·	•		1:	- 1 1	•	1	$1 \overline{2}$
$-\frac{14}{15}$ *	1 3 000 May 3000	-	-	and the same of th	$-\frac{2}{3}-\frac{2}{3}$	<u> 2 _ 1</u>	4_	37_
16		Т.		1 1	2 6	9 7	3	5 8
17		•		i	5 6	2 1 3 1	9	8 13 7 16
\frac{18}{19**} -		1	1 1	$-3 - \frac{2}{2}$	6 4	ĭ	11	
			3	<u></u>	16 10	3	19	T5 34
20 21		. 1	9 T	1 2 1 5	9 7	4 1 4 1	. 14	11 25
22	ן 'ד	2	3 1 2 3	2 2	13 4 17 5	4 1 7	21	11 32
23		1	ı	2 2 2 4	6 6	2 1	31 10	13 42 13 23
24	***	3	$\overline{1}$	ī	12 2	ĩ i	13	8 21
2 5		2	1 2	2	16 8	3	19	14 33
26 27		.	1 2	ļ		1	11.	8 19
28 28	r r	_		1 1	10 7	1	13	8 21
29	#	<u>.</u>	2 1	2 1 2	10 2	⊥	13 6	1 14 1 7
30		•	· · · · · · · · · · · · · · · · · · ·	4	4 i	١ .	5	1 6
31			1		4	•	5	5
32			, , , , , , , , , , , , , , , , , , ,		2		., Ž	2
			-					· ·
Totals	1 1	3 9	10 15	17 26	148 78	38 7	217	196 959
		· ·				. .	41	136 353

^{*}Necessary for admission of Mississippi students in 1963.

^{**}Necessary for admission of out-of-state students in 1962.

MEAN SCORES, STANDARD DEVIATIONS, AND CORRELATIONS FOR GRADE POINT AVERAGES OF 1961-62 ENTERING FRESHMEN RASED ON COMPOSITE STANDARD SCORES OF THE AMERICAN COLLEGE TEST, FOR 1961-62, THE UNIVERSITY OF MISSISSIPPI

1.51	1 1		`.	•						·	•	*				•		٠, •			•		•		·	
	(13)	.67	70	27.	. 82	.74	.72	.67	. 70	29	.78	72	.78	.70	.74	.72	80	62	တို့	99	. 66	50	.61	38	50	484
lative	(12)				1.03		1.46	1.71	*	¥			\\ .	· •			eç.	נטי	כיו		(Q)	ଫ	0	L.	2,96.1	1
Comp	(11)	10	!	18	20	16	27	20 20 20 20 20 20 20 20 20 20 20 20 20	61	62	20	105	ဓ	82	86	73	 69	26	45	37	22	<u>د</u>	12.	0	4	
ter	(10)	. 83	.61	.56	.84	.61	.72	•64	.73	.78	.77	.70	.85	.71	.77	.81	.80	. 64	.80	.78	.62	. 49	.67	47	1.33	448
1 1	(6)	1.46	1.78	1.37	1.30	1.70	1.58	1.88	• •		₩.	1.98	\		٠٧ •	S	GD.	4	t.		တ	4	ರಾ	ໜ້	7	l
2nd	(8)	7	S	15	16	12	22	4	20	57	09	95	76	7.2	92	69	09	20	40	34	. 21	೮	12	∞	4	·
ster S n	(7)	63	1.12	. 58	. 74	84	.75	99.	.72	က ထ	08.	8 4	6/.	.77	80	08	- 81	. 79	80	. 79	.79	19		322		497
t Semes	(9)	88	•	1.08	86		1.47	•	•	٠ •	٠ •		ر ر •	٠ <u>.</u>	, K	· V		4.	4	ယ်ဖ	ဘ (3	o (Ö	3.06	
Is		10	ω	18	202	9 5	7.7) (2)	_ `	N (0	27.	105.	, XX (2 C	χ Ση	73	က ဖြ		45 5	%	77	უ.(- 1 /	77	œ	4	
ssion S.D.	(4)			(00°T	ο 6 Ο 6	٠ ا	n (D	ກ ຫຼື (ည်း သည်	0 0	23.	\$ \$ \$	() () ()	97.	0.0	٠ 0		1.5U		• 03			
ummer Se	(3)	2.00	(1,00	2. 2. 3. 6.	,€ '€	2		# ·	याः ५ स	יטעי מ)) Ç *		U	D L	ចិច	7 .	년.r	ન વ	ۍ د	4	•	/#· o		*** **	
Sur	(3)	H		 (7) G	, , , , , , , , , , , , , , , , , , ,	# O	0 0	ם ע	ِهُ الْ	0 <	# u) ר) 6 -	7	- 6	\ 6	Юu)	ન જ	?	•		-1 .		9
ACT		Below 10	3:	- G	\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	2.▼	# L		2.5	\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \) C		40	1 6 1 6	246	H U) 10 10 10 10 10 10 10 10 10 10 10 10 10	, c	i o	00) C) -		7	

MEAN SCORES, STANDARD DEVIATIONS, AND CORRELATIONS FOR GRADE POINT AVERAGES OF 1961-62 ENTERING FRESHMEN BASED ON COMPOSITE STANDARD SCORES OF THE AMERICAN COLLEGE TEST, FOR 1962-63, THE UNIVERSITY OF MISSISSIPPI

•	98	ľ			•				•	•	•	, ;	,			• (*					•		+				•					*	4		ľ	-
	Scor	8.0	(13)		6.		4	47	.67	100) () (. 28	54	47	۰	100	. 50	ָ נכ רבי	1 ×	40.*	67	99) (S	1 C	//:	.65	.74	62	22	, Y	# to 0	30	. 27	14	471	
	tive	ean	12)			•					. 7		7			•	Τ.		,,,	-	يّ	N.A	7	. 6.	7	ri y	6.7	CO	Q;	, è	J r	4	9	95 1	!	
. ·	Cumula	0)	(4	σ	•	ņ		. 9) 	- ∤ (ゔ	Š	i	•	, H	മാ	~) G	.		-44					-	ა ი	4 2.		a
	ပ	N	ز				•	•			,	_		u .,		,	4			7		<u></u>	, LC	7	μ,	4	က	ന	20	ر ا	1 ~	7		1		
		S.D.	(10)		. 56	63	1 5	// *	.47	.87	, o	00	98.	.53	VA	H (• 66	09.	78) C	8/.	.75	.81	8	1,0	//.	96.	. 75	.81	5.			4.	1.66	.378	
1	nd Semes	Mean	(6)		a	-	•	п		_	, ~	•	•	•		•	_	v	,07	įr	7	w)	4	u,	ט פ	ו כו	Q		∞		9	1.	3.47	eð.		
ŀ	27	S _N	(8)		دی	ଦ୍ର) C	- !	<u> </u>	©	19	1 6 1 6	/7	လ 4	٠ ٢	3 (4	4 0	63	49	0 0	0	63	48	43) (c	\ 0	N .	300	13	11) L	က (n		
	7-1	-;-		,	Ţ	ಟ	7	# :	_	<u>8</u>	<u> </u>	2 (O I		N.	· •	2 (ව	ເນ	·		>	တ	~	ď	.		י כים	ထ	4	0) ((Ö 0		4.	
	Ste	S	٦	•	•	•		•	•		٠,	,	•	•	ω.		•	9.		α	2 (1		1.0	α	•) (Φ.			. 7	· 6	1		, M	•
K	T Vene	Mear	9		. 🛛		3	•				,	•	*	٦.	_	- `		~		15	Ä,		oĀ	7	-1 د	~ C	Ų (9		Ø	C) () () ()	ת		٠
	ST.	SI SI	0						٠	∞	4	٥	t Dic	ا ۵	<u> </u>	σ	L	റ	N	سندم	4 6	5 (M		~	`		.		•			> 4			•
			,			,							*		-		*						,					•					१ १३			
	TO C		1#1	e n	. 60	S S	.87	. G	40.	40	က	α) (d	000	T-02	06	40	2	4. 4.	0	64	2 0	70.	က ဘ	. 74	œ œ) (I	00.	4/•	(r. 00	٠.		209	7	
E s	T CE	Die	101	•			્રે '∎	, –		i		٠,		"		٧٧	-	ر -	ا ب	~~	7	? (V	. J. C	טַ עַ	S			, C) (1.0	\supset	\Box	00) :		
Z.I.mm	NO CENTRAL		16/			1							् •	1 c) (œ	4	H 0	O '	4	σ	F 	7 G H L	- 1 (つ (- (N)	7	1.2	i . I		
				.			j	18-23 18-23 18-24 18-24		_ \$			ر ا ا ا ا د ا	. t.			F.			4. 4.	• 7					# ·		چ آ	3	- -	φ. =					
	y. Q			٦ ا		- ء د	<u> </u>	∾	6º) •	.	ر م	· ·			~	<u>C</u>		· ·			وندو				-	 		, ,				****			
	Scor	F	1	Bello) 	1 r	— ાં જિ			Î-	1					-		Š	3 6	7	Ċ,	4		7 G	Ý	2	ં	6)	3 ¢) ()	٦ ص	හ භ	H	6 U	

TABLE V-3-C

MEAN SCORES, STANDARD DEVIATIONS, AND CORRELATIONS FOR GRADE

	lative Sc	(12) (13)		1.95 .47	1.46	1.44	1.92 .1	1.78.	2.20	2.06 .5	2.06 .5	2.06.5	2.22 .5	2.32 .6	2.37 .5	2.37 .5	2.60 .5	2.646	2.61 .5	2.55.6	2.87 .5	2.91 .6	3.20 .5	3.15	3,55 .3	2.94 .9	492
RING FRESHMEN BASED ON THE AMERICAN COLLEGE ERSITY OF MISSISSIPPI	nd Semester	(8) (9) (10) (11		3 1. 75 .64 3	1.56 ,52	1,49 .52	4 2.42 .20	3 2.19 .61 1	9 2.54 .46 2	9 2.24 .76 2	2.24 .68	1 2.34 .73	4 2.40 .86 6	8 2.65 .60	8 2.68 .56 5	3 2.50 .79 6	9 2.79 .72	8 2.76 .72	8 2.74 .65	4 2.64 .86	5 3.05 .57 2	5 2.98 .76 1	3.28 .60	3.20 .67	3.29.59	3.18 .65	.352
RAGES OF 1961-62 ENTE TE STANDARD SCORES OF FOR 1963-64, THE UNIV	t Semest	(5) (6) (7)	7 03 1	2 - 1 - 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 -	1.71	1.6	4 2.43	3 1.78	2 2.16	7 2.24	2.17	0 2.13 U	7. 2.25	2 2.52 .7	9 2:53	3 2.56	9 Z9 Z	7 Z 84	2.64	1 2.74	3.04	2.85	9. U.S.	e Te	3.48 3.	0. Z.O	C14*
POINT AVE COMPOSI TEST,		2) (3)			•	∾.	ω	2Z	N. 0	7. 2.67	2/.2	27.7°7°	0	7 44 0	C. 24.2 5	/* 0/*Z T	4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4		7. 70.7	0. 48.44 0. 6. 6. 6.	5. 21.0	7. 50.7	C	7 C		•	0
	ACI			3 07	H.	M	n	41	(A)	10 H	\ \ \	0< 0 1 -	n c) r	-1 G	790	2 d	# W 9 ©) U	2 C	~ O	0 G V 6	7 G) (100		

TABLE V-3-D

MEAN SCORES, STANDARD DEVIATIONS, AND CORRELATIONS FOR GRADE POINT AVERAGES OF 1961-62 ENTERING FRESHMEN BASED ON COMPOSITE STANDARD SCORES OF THE AMERICAN COLLEGE TEST, FOR 1964-65, THE UNIVERSITY OF MISSISSIPPI

ž,			·	. .	•					•		•		.'		_	
Scores S.D. (13)	.71		င ်း နှံ့	48 52	. 57 . 46	225	4 4 ር	. 52	1.03	5.4 4.4	52		192	40	رة الم	489	
Mean (12)	L.6.	1.64.1	ന <u>ല</u>	2.2		ינים ו עם (2.02	43 c	7 6	7.7	0, 0	vi ec	က	က ရ	2	,
Cumuil No.,	4.	မ လ လ	125	17	8 C	4.	34	52	4.0 4.0	35 30	. 20	12.	- 4, 다.	2	4.6	7	
Ster S.D. (10)	.40	 	たる もな4	.75	27.	74	80.5		.73		36	.52	T/.		.78	348 348	
Sеще Mean (9)	88 81 1-1	G	7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	900	102 c	7 01 4 0	2,0	N 60	\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\	ω °	2 C	က မ	∞2 α	၁ က ၁ က	8	ຕິ	
2nd No. (8)	က	41 ¢	4-	2 2	17	35 35	24	24 e.	23	26	4 C	18	တ	ာ ဖ	∀ †	~ 3	
Ster S.D. (7)	89.		900			7.7		ω. 4. α	08	. 78	 	09	60°	L 23		1.06	!]
st Seme	an H		2 2 2 2 2 2 2 2	<u>.</u> สูญ ค	3 63 6 4 Ci 6	ഗ ഡ് 4	2.5	20 c 40 c	2 67 5 CJ	8	22 c	4 4	%	7 C	် က	87	
	7		, , , , , , , , , , , , , , , , , , ,	1 - 1 6	4 67	X	က	केंद्र र	# 63	က	% -	-1 07	H			ا ان	
ession n S.D. (4)	.25.	ଷ୍ଟ :	ស •	# L	ກືຜູ	∞ Γ		Φ.	, 0		ထွမ	n (.77	1	ר כ
mmer S Mea	2.	rd (യ പ യ പ യ പ	4 M	2 Z 2 Z 2 Z	00 C 2. €) W	22.2	~ ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	8 2 2	∞ 6	ω « Μ	, cd	കാ ദ	ည် ဆုံ သူ ရာ		
						H §			7 (ान्तु । जिल्हें जिल्हें	11	ا المستشرسية المستشرسية				
ACT Scores	Below 1		이 # H 규	4 C	1 <u>0</u>	Φ C	7 Q	S	70	1 01 3 01	1 0	26	× &	တ လ		N	

Chapter VI

SUMMARY AND CONCLUSIONS

I. SUMMARY

At the beginning of the school session 1961-62 there were 1062 students (669 men and 393 women) who enrolled in the freshman class at The University of Mississippi. group was the first required to take the American College Test when entering the University, but no minimum score was required for admission. The range of scores extended from a low composite standard score of four, which ranked below the first percentile for entering students, to a high composite standard score of 32, which ranked above the ninety-ninth Four students scored a composite standard score percentile. of 32 and the median composite standard score for the 1961-62 entering freshmen was 21. In the individual subject areas the ranges of scores and the medians of scores were as follows: English, one to 36 with a median score of 21; mathematics, two to 36 with a median score of 20; social studies, one to 33 with a median score of 21; and natural science, one to 34 with a median score of 21.

In 1962 the Board of Trustees of State Institutions of Higher Learning authorized any institution under its jurisdiction that desired to do so, to set a minimum score for admission purposes. The University of Mississippi set a minimum composite standard score of 15 for resident students of the state and a minimum of 19 for non-resident students. Thus the 1961-62 entering freshmen used in this study was the

For out-of-state students. The restriction for in-state students was delayed until 1963-64. These score limits were used in this study to analyze the academic success of three groups; those with ACT scores of less than 15, those with scores of 15-18, and those with scores of 19 and above. Enrollments in three groups were: those with scores of less than 15--102, 7 per cent; 15-18--250, or 23.5 per cent; 19 and above, 709, 31 66.8 per cent.

mean grade into which this study was divided, the mean grade int averages and standard deviations by each individual composite standard score were determined. Correlation coefficients between ACT scores and grade point averages of scores in each group were determined for each semester and each summer sellon except the 1961 summer session.

Also, similar coefficients of correlation were found between the scores and the cumulative or ade point averages.

For the group of students with composite standard scores of less than 15 on the American College Test, correlations between their test scores and their grade point averages for a single semester or summer session ranged into -.311 during the summer session of 1962 and the third semester, respectively, to .295 and .259 during the sixth and eighth semesters, respectively. The correlation was near zero for five periods of enrollment. Also, for the group the correlation coefficients between the cumulative grade point averages and composite standard scores on the test were as follows: first year, .157; second year, *.072; third year, -.062; and fourth year, .161.

For the students with composite standard scores of 15, 16, 17, 18, the correlation coefficients ranged from -.070 during the summer session of 1963 to a correlation of .096 during the third semester of enrollment. The correlation coefficients for every enrollment period indicate little relationship between scores of 15, 16, 17, 18 on the American College Test and grade point averages. For this group of students the cumulative correlation coefficients, determined at the end of each academic year, were: first year, .031; second year, .006; third year, -.076; and fourth year, -.047. The third group of students (with composite standard scores of greater than 18 on the American College Test) showed correlation coefficients between scores and grade point averages that ranged from .140 during the summer session of 1965 to high correlations of .378 and .364 during the second and first semesters, respectively. The correlation coefficients between the test scores and the cumulative grade point averages determined at the end of each academic year were; first year, .385; second year, .387; third year, .387; and fourth year, .357.

Correlation coefficients between the cumulative grade point averages and the composite standard scores of the American College Test for all of the students were: first year, .484; second year, .471; third year, .492; and fourth year, .489.

Attrition in any class needs careful examination and this has been done, by semesters, in this study. During the eighth semester of enrollment there were only 340 of the 1062

students of the class under study remaining at The University of Mississippi. Table VI-1 shows a final summary of the students by indicating the disposition of the students who were included in the 1961-62 entering freshman class. Disposition of these students are classified in four ways: those who graduated by the end of the 1965 summer session; those who were enrolled at The University of Mississippi at the end of the second semester of the 1964-65 academic year or the 1965 summer semester and who had not graduated; transfer students (includes all students who requested transcripts be sent to other institutions whether or not they actually enrolled in another institution); and the students who were not included in the three preceding catagories who were classified as dropouts.

TABLE VI-1
DISPOSITION OF THE 1961-62 ENTERING FRESHMEN
BY SEX, ON AUGUST 22, 1965, THE
UNIVERSITY OF MISSISSIPPI

	Number			Per Cent		
Disposition	Men	Women	Total	Men	Women	Total
(1)	(2)	(3)	(4)	(5)	(6)	(7)
Graduated	217	136	353	32.4	34.6	33.2
Enrolled*	139	18	164	20.8	4.6	15.4
Transferred	209	123	325	31.2	31.3	30.6
Dropped	104	116	220	15.5	29.5	20.7
Totals	669	393	1062	100.0	100.0	100.0

Note: Per cents do not total 100 due to rounding.

*Enrolled in either or both the spring semester or summer session 1965 but had not graduated by the end of the 1965 summer session.



II. CONCLUSIONS

From the study of the academic achievement of the 1961-62 entering freshmen based on the composite standard scores earned on the American College Test the following conclusions were reached:

- ŀ. When all scores are considered for the entire freshman group, the correlation coefficients between the composite standard scores and the academic achievement are positive, both on a semester by semester basis and also on the basis of cumulative grade point averages. The same is true, although considerably lower for the group with scores of above 18. the other two groups, the one with scores below 15 and the one with scores of 15 through 18, both positive and negative coefficients of correlations were found and all were quite small. On a semester basis, coefficients of correlation between scores and grade point averages were sometimes positive and sometimes There were a number of students who were quite acanegative. demically capable, but who did not score high on the American College Test.
- 2. The score of 19 seems to be the lowest score that is necessary to produce a mean passing grade in the very first and subsequent semesters of college.
- 3. Additional tests, as well as some that are of a different type from the ACT, should be given to foreign students who apply for admission to The University of Mississippi,



if academic ability is to be determined. 1

- 4. The dropout rate is inconsistent from ACT score to ACT score. Dropouts were found for every score except score 31 where five of eight studerts graduated and the other three asked for transcripts to other schools. The greatest change of dropouts was between the scores of 20 and 21.
- 5. Approximately one third of the 1961-62 entering freshmen had graduated from The University of Mississippi at the end of the 1965 summer session following their fourth college year and about 29 per cent had graduated in June 1965 at the end of the traditional four year period. Summer sessions played an important part for many of the students in this class.
- 6. It is quite possible that an additional 15.4 per cent of the class, enrolled in the 1965 spring semester or summer session but not graduated, will graduate. Another 30.6 per cent of the students who had transcripts of their credits sent to other schools, will have some from their number to graduate. Thus, there is a possibility of approximately

¹Three foreign students scored less than 15 as a composite standard score on the American College Test. these was a young lady from Nicaragua who made a composite standard score of five and was enrolled in the School of Business and Government. She graduated in June, 1965, with a cumulative grade point average of 2.23. A student from Syria, who had a composite standard score of 11, enrolled in the School of Engineering. He transferred to the University of Michigan at the end of the 1963 summer session with a cumulative grade point average of 2.58. A third foreign student, who was from Hong Kong, made a composite standard score of He enrolled in the School of Engineering. When he transferred to San Diego State College at the end of the third semester, he had a grade point average of 3.29. The lowest individual subject score for each of these students on the American College Test was in the area of English. The test does not seem to identify potentially able students from foreign countries.

- 79.0 per cent of the students completing requirements for a degree either at The University or at other institutions.
- 7. The American College Test serves as a good indicator of academic success for entering freshmen. However, there are some students who score low on the test that are able to achieve very well. On the other hand, consideration should be given to finding a method to reduce the number of dropouts among the students with high ACT scores.